D OF SUPERVISORS T CONTENTS LIST
mmittee Date
Analyst Report ort er Letter and/or Report nission
nal space is needed)

_Date __ _Date__

Completed by: Linda Wong
Completed by: Linda Wong

[Administrative Code - Establish Low Carbon Fuel Standard Credits Sales Fund - Authorizing Agreements to Sell Low Carbon Fuel Standard Credits - Public Utilities Commission]

Ordinance delegating authority under Charter, Section 9.118(a), to the General Manager of the Public Utilities Commission to enter into one or more agreements to sell Low Carbon Fuel Standard Credits accrued by the Municipal Transportation Agency when revenue from such an agreement will be \$1,000,000 or more; and amending the Administrative Code to establish a Low Carbon Fuel Standard Credits Sales Fund to receive funds from the credits sales for use in connection with projects that will reduce San Francisco's carbon footprint and/or advance its sustainability goals.

Unchanged Code text and uncodified text are in plain Arial font.

Additions to Codes are in single-underline italics Times New Roman font.

Deletions to Codes are in strikethrough italics Times New Roman font.

Board amendment additions are in double-underlined Arial font.

Board amendment deletions are in strikethrough Arial font.

Asterisks (* * * *) indicate the omission of unchanged Code subsections or parts of tables.

Be it ordained by the People of the City and County of San Francisco:

Section 1. Findings.

NOTE:

- (a) In Assembly Bill 32, entitled the Global Warming Solutions Act of 2006, the Legislature required California to reduce its greenhouse gas emissions to 1990 levels by 2020.
- (b) In 2009, at the direction of then-Governor Schwarzenegger, the California Air Resources Board ("CARB") adopted Low Carbon Fuel Standard ("LCFS") regulations to partially implement Assembly Bill 32 ("LCFS Regulations"). Under the LCFS Regulations,

transportation providers that use fuels with a carbon intensity (or greenhouse gas emissions) lower than the State target generate LCFS credits that can be sold.

- (c) The San Francisco Municipal Transportation Agency ("SFMTA") uses greenhouse gas-free electric energy provided by the San Francisco Public Utilities Commission ("SFPUC") to operate its electric public transit fleet, consisting of electric trolley buses, cable cars, light rail vehicles and, in the future, battery-powered electric buses. Because this fleet operates with a carbon intensity below the State target, the SFMTA accumulates credits that can be sold.
- (d) The SFPUC estimates that the LCFS credits it has accumulated to date can be sold for more than \$10 million at current prices. The SFPUC further estimates that, in subsequent years, it could accumulate credits worth in excess of \$4 million per year at current prices.
- (e) In October 2017, the SFPUC and the SFMTA entered into a Memorandum of Understanding ("MOU") setting forth the roles and responsibilities of the two agencies with respect to sales of the LCFS credits. Among other things, the agencies agreed that: (1) each agency would share equally the net revenues from the LCFS credits sales; and (2) both agencies would use the revenues generated from selling the LCFS credits to support general operations that reduce San Francisco's carbon footprint and/or advance the City's sustainability goals.
- (f) In Resolution No. 17-0199, dated September 12, 2017, the SFPUC approved the MOU, authorized the SFPUC's Power Enterprise to sell the LCFS credits, delegated to the SFPUC's General Manager the authority to enter into agreements necessary to effectuate those sales, and authorized the SFPUC to seek Board of Supervisors approval of any sales under the LCFS program that would exceed \$1 million.

(g) Pursuant to that authority, the SFPUC intends to sell the LCFS credits by issuing requests for offers to interested counterparties. Once the SFPUC accepts an offer from a counterparty, the SFPUC and the counterparty will enter into a Leadership for Energy Automated Processing ("LEAP") Master Agreement for Purchasing and Selling Low Carbon Fuel Standard Credits ("LEAP Master Agreement"). The LEAP Master Agreement will establish the necessary terms and conditions of the sale and the transfer of title to the LCFS credits from the SFMTA to the counterparty.

Section 2. Delegation of Authority Under Charter Section 9.118(a).

- (a) Under Section 9.118(a) of the Charter, the Board of Supervisors ("Board") must approve any contract entered into by a City department where the anticipated revenue from the contract is \$1 million or more.
- (b) The Board authorizes the General Manager of the SFPUC to sell the SFMTA's existing LCFS credits and future LCFS credits, through one or more contracts, including contracts with revenues of \$1 million or more. The Board is approving the SFPUC's entering into such contracts at this time, even before the SFPUC has offered the LCFS credits for sale, because the Board understands that the SFPUC will need to expeditiously complete the sales to maximize the value of the credits. In addition, the Board is granting this authority with the understanding that there could be multiple sales in excess of the \$1 million threshold (each of which would require Board approval absent this delegation), and that each sale will be effectuated through use of the LEAP Master Agreement.

1	
2	
3	
4	
5	
6	
7	
8	
9	
10	
11	
12	
13	
14	
15	
16	
17	
18	
19	
20	
21	
22	
23	
24	

(c) The Board conditions its approval of LCFS credits sales of more than \$1 million on the General Manager of the SFPUC entering into contracts substantially in the form of the LEAP Master Agreement; provided, however, that the City Attorney and SFPUC General Manager may determine that certain non-material changes to that form agreement are necessary to effectuate the LCFS credits sales authorized herein and/or to adequately protect the City's interests.

Section 3. Chapter 10, Article XIII, of the Administrative Code is hereby amended by adding Section 10.100-92, to read as follows:

SEC. 10.100-92. LOW CARBON FUEL STANDARD CREDITS SALES FUND.

- (a) Low Carbon Fuel Standard Credits Sales.
- (1) The San Francisco Municipal Transportation Agency ("SFMTA") uses greenhouse gas-free electric energy provided by the San Francisco Public Utilities Commission ("SFPUC") to operate its electric public transit fleet, consisting of electric trolley buses, cable cars, light rail vehicles, and, in the future, battery-powered electric buses.
- (2) In Assembly Bill 32, entitled the Global Warming Solutions Act of 2006, the Legislature required California to reduce its greenhouse gas emissions to 1990 levels by 2020. In 2009, at the direction of then-Governor Schwarzenegger, the California Air Resources Board adopted Low Carbon Fuel Standard ("LCFS") regulations to partially implement Assembly Bill 32 ("LCFS Regulations"). Under the LCFS Regulations, transportation providers that use fuels with a carbon intensity (or greenhouse gas emissions) lower than the State target generate LCFS credits that can be sold.

- (3) The SFMTA accumulates credits that can be sold to fuel producers, because the SFMTA operates its electric public transit fleet with a carbon intensity (or greenhouse gas emissions) below the State target. As of 2019, the SFPUC has agreed to take responsibility for the sales of those credits. It is possible that other agencies of the City and County of San Francisco might also accumulate LCFS credits and that the SFPUC might assume responsibility for the sales of those LCFS credits.
- (b) Establishment of Fund. The Low Carbon Fuel Standard Credits Sales Fund is established as a category eight fund to receive, and account for the expenditure of, proceeds from the sale of Low Carbon Fuel Standard credits by the SFPUC.

(c) Use of Fund.

- (1) Money in the Low Carbon Fuel Standard Credits Sales Fund shall be used first to reimburse the SFPUC and the SFMTA for any expenses they incur to sell Low Carbon Fuel Standard credits. After expenses, the SFPUC and the SFMTA will share equally in the net revenues from such sales. The SFPUC will use its share of the net revenues to support general operations that reduce the City and County of San Francisco's carbon footprint and/or advance its sustainability goals. The SFMTA intends to use its share to advance the SFMTA's sustainability and climate action initiatives and goals.
- (2) In the event that any other City agencies accumulate LCFS credits, and that the SFPUC takes responsibility for the sale of those LCSF credits, funds from those sales may also be received in the Low Carbon Fuel Standard Credits Sales Fund for distribution by the SFPUC to those other City agencies. In that event, the SFPUC and those other agencies will determine how to share the proceeds.

(d) Administration of Fund. The SFPUC will administer the Low Carbon Fuel Standard

Credits Sales Fund. The General Manager of the SFPUC shall submit an annual written report to the

Board of Supervisors and the Controller describing expenditures made from the Low Carbon Fuel

Standard Credits Sales Fund during the previous fiscal year.

Section 4. Effective Date. This ordinance shall become effective 30 days after enactment. Enactment occurs when the Mayor signs the ordinance, the Mayor returns the ordinance unsigned or does not sign the ordinance within ten days of receiving it, or the Board of Supervisors overrides the Mayor's veto of the ordinance.

APPROVED AS TO FORM: DENNIS J. HERRERA, City Attorney

Ву:

WILLIAM K. SANDERS Deputy City Attorney

n:\legana\as2019\1900494\01366001.docx

LEGISLATIVE DIGEST

[Administrative Code - Establish Low Carbon Fuel Standard Credits Sales Fund - Authorizing Agreements to Sell Low Carbon Fuel Standard Credits - Public Utilities Commission]

Ordinance delegating authority under Charter, Section 9.118(a), to the General Manager of the Public Utilities Commission to enter into one or more agreements to sell Low Carbon Fuel Standard Credits accrued by the Municipal Transportation Agency when revenue from such an agreement will be \$1,000,000 or more; and amending the Administrative Code to establish a Low Carbon Fuel Standard Credits Sales Fund to receive funds from the credits sales for use in connection with projects that will reduce San Francisco's carbon footprint and/or advance its sustainability goals.

Existing Law

Section 9.118(b) of the San Francisco Charter requires approval by the Board of Supervisors for agreements with a City and County of San Francisco ("City") department or agency when revenue from such agreements will be \$1 million or more.

Amendments to Current Law

The proposed ordinance would amend the Administrative Code to establish the Low Carbon Fuel Standard Credits Sales Fund to receive the proceeds from any sales of Low Carbon Fuel Standard ("LCFS") credits by the San Francisco Public Utilities Commission ("SFPUC").

Background Information

In Assembly Bill 32, entitled the Global Warming Solutions Act of 2006, the Legislature required California to reduce its greenhouse gas to 1990 levels by 2020.

In 2009, at the direction of then Governor Schwarzenegger, the California Air Resources Board ("CARB") adopted LCFS regulations to partially implement Assembly Bill 32 ("LCFS Regulations"). Under the LCFS Regulations, transportation providers that use fuels with a carbon intensity (or greenhouse gas emissions) lower than the target established by CARB generate LCFS credits that can be sold.

The San Francisco Municipal Transportation Agency ("SFMTA") uses greenhouse gas-free electric energy provided by the SFPUC to operate its electric public transit fleet (electric trolley buses, cable cars and light rail vehicles). Because this fleet operates with a carbon intensity below the State target, the SFMTA accumulates credits that can be sold.

The SFPUC estimates that the LCFS credits it has accumulated to date can be sold for more \$10 million at current prices. The SFPUC further estimates that, in subsequent years, it could accumulate credits worth in excess of \$4 million per year at current prices.

In October 2017, the SFPUC and the SFMTA entered into a Memorandum of Understanding ("MOU") setting forth the roles and responsibilities of the two agencies with respect to sales of the LCFS credits. Among other things, the agencies agreed that they would: (i) share equally the net revenues from the LCFS credits sales; and (ii) use the revenues generated from selling the LCFS credits to support general operations that reduce San Francisco's carbon footprint and/or advance the City's sustainability goals.

In Resolution No. 17-0199 dated September 12, 2017, the SFPUC approved the MOU, authorized the SFPUC's Power Enterprise to sell the LCSF credits, delegated to the SFPUC's General Manager the authority to enter agreements necessary to effectuate those sales, and authorized the SFPUC to seek Board of Supervisors approval of any sales under the LCFS program that would exceed \$1 million.

The proposed ordinance would authorize the SFPUC to sell the SFMTA's LCFS credits, through one or more contracts, including contracts with revenues of \$1 million or more.

The proposed ordinance would also amend the Administrative Code to establish the Low Carbon Fuel Standard Credits Sales Fund as a category eight fund to receive, and account for the expenditure of the proceeds, from the LCFS credits sales. The SFPUC will be responsible for administering the fund and distributing the net revenues in accordance with the MOU.

n:\legana\as2019\1900494\01365831.docx

Item 1	Department:
File 19-0656	Public Utilities Commission (PUC)

EXECUTIVE SUMMARY

Legislative Objectives

- Charter Section 9.118(a) requires Board of Supervisors approval for agreements in which
 revenues are anticipated to be \$1 million or more. The proposed ordinance would waive
 this Charter requirement, delegating authority to the San Francisco Public Utilities
 Commission SFPUC General Manager to enter into one or more agreements to sell Low
 Carbon Fuel Standard Credits accrued by San Francisco Municipal Transportation Agency
 SFMTA when revenue from such agreements equals \$1 million or more.
- The proposed ordinance also establishes a Low Carbon Standard Credits Sales Fund to receive funds from the credit sales.

Key Points

- SFMTA uses electricity supplied by the San Francisco Public Utilities Commission (SFPUC) to operate the electric trolley buses, cable cars, and light rail. All of this electricity comes from hydropower generated by Hetch Hetchy Water and Power, and therefore, is a zero-carbon energy source.
- The California Air Resources Board adopted Low Carbon Fuel Standards in 2009, which set low carbon fuel targets for transportation providers. Providers with fuel use below these targets can receive Low Carbon Fuel Standard Credits which they can then sell.

Fiscal Impact

• SFPUC has not previously sold Low Carbon Fuel Standard Credits. According to SFPUC, SFMTA accumulated approximately 18,000 credits per year between Calendar Year (CY) 2016 and CY 2018 or 54,000 credits over the three-year period. Based on an average price of \$160 per credit in CY 2018, the 54,000 credits would be equal to approximately \$8.6 million. Based on the average price of \$186 for the first two quarters of CY 2019, the 54,000 credits would be equal to approximately \$10.0 million.

Policy Consideration

• Approval of the proposed ordinance is a policy matter for the Board of Supervisors because it would delegate the Board's authority under Charter Section 9.118(a) to the SFPUC General Manager to enter agreements to sell Low Carbon Fuel Standard Credits accrued by SFMTA when revenue from such agreements equals \$1 million or more. If the Board of Supervisors approves the proposed ordinance, the Board should request the SFPUC to (a) report on the SFPUC's protocols for selling Low Carbon Fuel Standard Credits, and (b) report annually on the sale of Low Carbon Fuel Standard Credits, including the number sold, the price per credit, and the total amount of the sales

Recommendations

- Approval of the proposed ordinance is a policy matter for the Board of Supervisors.
- If the Board of Supervisors approves the proposed ordinance, the Board should request the SFPUC to (a) report on the SFPUC's protocols for selling Low Carbon Fuel Standard Credits, and (b) report annually on the sale of Low Carbon Fuel Standard Credits, including the number sold, the price per credit, and the total amount of the sales.

MANDATE STATEMENT

City Charter Section 9.118(a) states that any contract entered into by a department, board or commission that has anticipated revenues of \$1 million or more is subject to Board of Supervisors approval.

BACKGROUND

The San Francisco Municipal Transportation Agency (SFMTA) uses electricity supplied by the San Francisco Public Utilities Commission (SFPUC) to operate the electric trolley buses, cable cars, and light rail. All of the electricity supplied by SFPUC to the SFMTA comes from hydropower generated by Hetch Hetchy Water and Power, and therefore, is a zero-carbon energy source.

The California Air Resources Board adopted Low Carbon Fuel Standards in 2009, which set low carbon fuel targets for transportation providers. Transportation providers using fuel below the low carbon fuel targets can receive Low Carbon Fuel Standard Credits which they can then sell.

Memorandum of Understanding (MOU)

SFMTA registered with the Low Carbon Fuel Standards Program in September 2016 soon after the program became fully operational and began accumulating Low Carbon Fuel Standard Credits. SFMTA and SFPUC entered into an MOU in 2017 in which the SFPUC would sell the credits on behalf of the SFPUC and SFMTA with proceeds used to fund SFMTA sustainability and climate action initiatives, and SFPUC clean energy projects. The proceeds from the sale of the Low Carbon Fuel Standard Credits were to be shared equally between the two agencies.

Low Carbon Fuel Standard Credits

Each Low Carbon Fuel Standard Credit is equivalent to one ton of CO_2 emissions. The price per carbon credit has increased since 2017. The average price per Low Carbon Fuel Standard Credit and total number of Low Carbon Fuel Standard Credits transacted in California are shown below.

Calendar Year	Total Number of Credits	Average Price per Credit
CY 2016	5,343,000	\$101
CY 2017	8,875,000	\$89
CY 2018	13,334,000	\$160

Source: California Air Resources Board Monthly Low Carbon Fuel Standard Credit Transfer Activity Report

The average price per credit for CY2019 Q1 and Q2 has further increased to \$186.

DETAILS OF PROPOSED LEGISLATION

Charter Section 9.118(a) requires Board of Supervisors approval for agreements in which revenues are anticipated to be \$1 million or more. The proposed ordinance would waive this Charter requirement, delegating authority to the SFPUC General Manager to enter into one or more agreements to sell Low Carbon Fuel Standard Credits accrued by SFMTA when revenue from such agreements equals \$1 million or more.

The proposed ordinance also establishes a Low Carbon Standard Credits Sales Fund to receive funds from the credit sales. Funds would be allocated equally to SFMTA and SFPUC in accordance with the 2017 MOU to be used for projects and operations that reduce the City's carbon footprint.

Process for Selling Low Carbon Fuel Standard Credits

SFPUC would sell carbon credits in batches through a Request for Bid process. All bidders must be registered with the California Air Resources Board, which tracks all Low Carbon Fuel Standard Credits and serves as a tracking/clearinghouse for carbon credit transactions. The transaction would be formalized through the LEAP¹ Low Carbon Fuel Master Agreement, which standardizes the terms, definitions, and other provisions of the transaction. The amount and price of all Low Carbon Fuel Standard Credit transactions are reported to the California Air Resources Board and publicly posted on the Board's web-site.

According to SFPUC staff, the SFPUC would monitor price and volume data reported on the California Air Resources Board website to determine optimal sales times. SFPUC would also develop protocols as needed, such as setting a minimum price and number of bidders for sales transactions, and reserving the right to not proceed with sales if the price is not sufficient. The sale of Low Carbon Fuel Standard Credits would be managed by the SFPUC's procurement group, which also manages electricity market transactions.

FISCAL IMPACT

SFPUC has not previously sold Low Carbon Fuel Standard Credits. According to SFPUC, SFMTA accumulated approximately 18,000 credits per year between Calendar Year (CY) 2016 and CY 2018 or 54,000 credits over the three-year period. Based on an average price of \$160 per credit in CY 2018, the 54,000 credits would be equal to approximately \$8.6 million. Based on the average price of \$186 for the first two quarters of CY 2019, the 54,000 credits would be equal to approximately \$10.0 million.

According to SFPUC staff, SFMTA is expected to accumulate an additional 25,000 (approximate) Low Carbon Fuel Standard credits in CY 2019, based on the SFPUC's documentation that SFMTA's energy needs are being served by zero-greenhouse gas renewable energy that meets California's renewable portfolio standards.

POLICY CONSIDERATION

Approval of the proposed ordinance is a policy matter for the Board of Supervisors because it would delegate the Board's authority under Charter Section 9.118(a) to the SFPUC General Manager to enter agreements to sell Low Carbon Fuel Standard Credits accrued by SFMTA when revenue from such agreements equals \$1 million or more. If the Board of Supervisors approves the proposed ordinance, the Board should request the SFPUC to (a) report on the SFPUC's protocols for selling Low Carbon Fuel Standard Credits, and (b) report annually on the

¹ "Leadership for Energy Automated Processing (LEAP)"

sale of Low Carbon Fuel Standard Credits, including the number sold, the price per credit, and the total amount of the sales.

RECOMMENDATIONS

- 1. Approval of the proposed ordinance is a policy matter for the Board of Supervisors.
- 2. If the Board of Supervisors approves the proposed ordinance, the Board should request the SFPUC to (a) report on the SFPUC's protocols for selling Low Carbon Fuel Standard Credits, and (b) report annually on the sale of Low Carbon Fuel Standard Credits, including the number sold, the price per credit, and the total amount of the sales.

Items 1, 2 and 3 Department:

Files 19-0356, 19-0357, 19-0358 | Port and Office of Public Finance

EXECUTIVE SUMMARY

Legislative Objectives

- **File 19-0357**: The proposed resolution would provide for the issuance of \$425,000,000 in Embarcadero Sewall Earthquake Safety General Obligation bonds to fund the first phase of reconstruction of the seawall.
- **File 19-0358**: The proposed resolution would authorize the sale of \$50,000,000 in General Obligations Bonds Series 2019B, and declare the City's intent to reimburse expenditures incurred prior to the issuance and sale of the Series 2019B Bonds.
- File 19-0356: The proposed ordinance would appropriate \$50,000,000 of the Series 2019B Embarcadero Seawall Earthquake Safety General Obligation Bond proceeds to the Port for planning, geotechnical, and other pre-development work, and place these funds on Controller's Reserve pending the sale of bonds.

Key Points

• In November 2018, San Francisco voters approved Proposition A, authorizing the City and County of San Francisco to issue up to \$425 million in to fund repairs and improvements to the Embarcadero Seawall and Embarcadero infrastructure and utilities for earthquake and flood safety. Phase 1 of the Embarcadero Seawall Program will be for immediate seismic and flood protection upgrades. The Phase 1 budget, including the \$425 million in General Obligation Bonds, is \$500 million.

Fiscal Impact

- Average annual debt service over 20 years on the Series 2019B GO Bonds is expected to be \$3,400,000. The estimated total principal and interest payment over the approximate 20-year life of the GO Bonds is \$68,300,000.
- If the Series 2019B GO Bonds are approved, the debt ratio would increase by 0.02 percentage points to 0.99 percent within the 3 percent legal limit.
- Debt service payments will be recovered through increases in the annual Property Tax rate, which, according to the Controller's Office, would \$1.32 per \$100,000 of assessed value. The owner of a residence with an assessed value of \$600,000, assuming a homeowner's exemption of \$7,000, would pay average additional property taxes to the City of approximately \$7.81 per year.

Policy Consideration

• On June 19, 2019, the Superior Court dismissed litigation regarding the Embarcadero Seawall Earthquake Safety General Obligation Bonds. However, because the 60-day period to file an appeal does not end until August 19, 2019, the Office of Public Finance does not plan to sell the bonds until after that date.

Recommendations

- Request the Office of Public Finance to submit a memorandum to the Board of Supervisors on the status of the potential appeal after the close of the 60-day period to appeal.
- Approve the proposed ordinance and resolutions.

MANDATE STATEMENT

City Charter Section 9.105 provides that the issuance and sale of General Obligation (GO) bonds is subject to Board of Supervisors approval in accordance with State law or local procedures adopted by ordinance.

City Charter Section 9.105 states that amendments to the Annual Appropriation Ordinance are subject to Board of Supervisors approval by ordinance after the Controller certifies the availability of funds.

BACKGROUND

The Embarcadero Seawall, which is over 100 years old, supports San Francisco's piers, wharves, businesses, tourist destinations, recreational amenities, and key infrastructure, including Bay Area Rapid Transit, Muni Metro, and ferry networks. The Embarcadero Seawall is the foundation of three miles of the city's waterfront, stretching from Mission Creek to Fisherman's Wharf. The Seawall also provides flood protection to downtown San Francisco.

In November 2018, San Francisco voters approved Proposition A, authorizing the City and County of San Francisco to issue up to \$425 million in bonds at an estimated tax rate of \$0.013 per \$100 of assessed value to fund repairs and improvements to the Embarcadero Seawall and Embarcadero infrastructure and utilities for earthquake and flood safety.

The first bond sale is expected to occur in June 2019 upon approval of File 19-0358. Table 1 below shows the proposed timeline for the total \$425,000,000 in bond sales.

Table 1. Embardació Scawan and infrastracture do Bona Sale Schedale.				
	Date	Amount	Series	
Proposed First Bond Issuance	FY 2018-19	50,000,000	Series 2019B	
Future Second Bond Issuance	FY 2021-22	250,000,000	•	
Future Third Bond Issuance	FY 2023-24	125,000,000		
Total Bond Issuances		425,000,000	·	

Table 1: Embarcadero Seawall and Infrastructure GO Bond Sale Schedule

DETAILS OF PROPOSED LEGISLATION

File 19-0357: The proposed resolution would (a) provide for the issuance of not to exceed \$425,000,000 aggregate principal amount of City and County of San Francisco General Obligation Bonds; (b) authorize the issuance and sale of the bonds; (c) provide for the levy of a tax to pay the principal and interest of the bonds; (d) provide for the appointment of depositories and other agents; (e) adopt findings under the California Environmental Quality Act (CEQA), and San Francisco Administrative Code Chapter 31; (f) find that the proposed project is in conformity with the priority policies of Planning Code Section 101.1(8) and with the General Plan consistency requirement of Charter Section 4.105 and Administrative Code 2A.53; (g) ratify certain actions previously taken; and (h) grant general authority to City officials to take necessary actions in connection with the issuance and sale of said bonds.

File 19-0358: The proposed resolution would (a) authorize the issuance and sale of not to exceed \$50,000,000 aggregate principal amount of City and County of San Francisco General Obligations Bonds Series 2019B, (b) prescribe the form and term of such bonds; (c) provide for the appointment of depositories and other agents for the bonds; (d) provide for the establishment of accounts related to the bonds, (e) authorize the sale of the bonds by competitive or negotiated sale, (f) approve the forms of the Official Notice of Sale and Intention to Sell Bonds, (g) approve the purchase contract, (h) approve the form of Preliminary Official Statement and the execution of the Continuing Disclosure Certificate, (i) approve the form of the continuing disclosure certificate, (j) authorize and approve modifications to such documents, (k) ratify actions taken previously, and (l) grant general authority to City officials to take necessary actions in connection with the authorization, issuance, sale and delivery of bonds.

In addition, the proposed resolution would declare the City's intent to reimburse expenditures incurred prior to the issuance and sale of the Series 2019B Bonds in connection with the project or portions thereof.

File 19-0356: The proposed ordinance would appropriate \$50,000,000 of the Series 2019B Embarcadero Sewall Earthquake Safety General Obligation Bond Proceeds to the Port of San Francisco in FY 2018-19 for planning, site, and geotechnical investigations, risk assessment, and other pre-development work, and placing these funds on Controller's Reserve pending the sale of bonds.

The proposed resolution (File 19-0358) ratifies all actions previously taken for the \$50 million in Series 2019B Embarcadero Sewall Earthquake Safety General Obligation Bond Proceeds. According to Mr. Mark Blake at the City Attorney's office, these actions include engaging bond counsel, the hiring of municipal advisors by the Controller's Office, and Port staff work.

Table 2 below outlines anticipated sources and uses for the bonds.

Table 2: Sources and Uses of Series 2019B Bond Proceeds

Sources	•
Par Amount	\$47,010,000
Reserve Proceeds	2,990,000
Total Sources	\$50,000,000
Uses	
Administrative Costs*	
Costs of Issuance	\$601,290
Underwriter's Discount	470,100
Controller's Audit Fund (0.2%)	91,600
Citizens' GO Bond Oversight Committee	47,010
Reserve for Market Uncertainty	2,990,000
Administrative Costs Subtotal	\$4,200,000
<u>Projects</u>	
United States Corp of Engineers Flood Study	\$8,875,000
Planning	14,965,000
Preliminary Design and Approvals, Phase 1	15,070,000
Regulatory Approvals	983,000
Program Management	5,907,000
Projects Subtotal	\$45,800,000
Total Uses	\$50,000,000

^{*}Numbers are based on estimate provided by Urban Future Inc., a municipal advisory firm registered with the municipal Securities Rulemaking Board (MSRB)

According to Mr. Trivedi, the Port has not yet allocated the \$375 million remaining bond proceeds. The first bond issuance will go towards planning, geotechnical investigations, and preliminary design and approvals. The Port plans to allocate the remaining bond proceeds based on the results from the initial planning, investigation, and preliminary design.

Phase 1 Project Costs

Phase 1 of the Embarcadero Seawall Program will be for immediate seismic and flood protection upgrades. The original budget for Phase 1, including the \$425 million in General Obligation Bonds, is \$500 million. As of April 18, 2019, the Port has secured \$446 million of the \$500 million, as shown in Table 3 below.

Funding Sources	FY 2016-17	FY 2017-18	FY 2018-19	FY 2019-20	FY 2021-22	FYs 2023-27	Total
Port Capital	\$2,900,000		\$1,100,000			10,000,000	\$14,000,000
City Revolving Fund	1,000,000	3,000,000	5,000,000	(6,000,000)	(3,000,000)		\$0.
SFMTA Contribution	500,000	500,000					\$1,000,000
Planning Department	-500,000	250,000	250,000				\$1,000,000
State Sources			5,000,000				\$5,000,000
2018 GO Bond (File 19-0356-8)	•		50,000,000		250,000,000	125,000,000	\$425,000,000
Gap		,				54,000,000	\$54,000,000
Total Planned Sources	\$4,900,000	\$3,750,000	\$61,350,000	(\$6,000,000)	\$247,000,000	\$189,000,000	\$500,000,000

The Port plans to fund the gap of \$54 million through a combination of State Resilience Bonds, state cap and trade revenue, state general fund budget requests, and development of a Mello-Roos District¹ along the waterfront. Phase 1 is expected to be completed by 2026.

The entire project consists of three phases, and will cost approximately \$5 billion in total.

FISCAL IMPACT

Annual Debt Service

As shown above in Table 2, the Office of Public Finance expects to sell \$47,010,000 in par value Series 2019B bonds.

The proposed resolution authorizes the Director of Public Finance to determine the sale date, interest rates, principal amount, and maturity dates of the bonds, subject to the following conditions: (1) the true interest cost shall not exceed 12 percent; and (2) the maturity date shall not be after thirty years of their date of issuance, approximately June 15, 2039.

The Office of Public Finance estimates that, based on a conservative estimate of 3.94 percent interest rate, that the average annual debt service over 20 years on the Series 2019B GO Bonds is \$3,400,000. The anticipated par value of \$47,010,000 is estimated to result in approximately \$21,700,000 in interest payments over the 20 year life of the bond. The estimated total principal and interest payment over the approximate 20-year life of the GO Bonds is \$68,300,000, of which \$21,700,000 is interest and \$47,010,000 is principal.

¹ Mello-Roos is a form of financing that can be used by cities, counties, and special districts (such as school districts). Mello-Roos Community Facilities Districts (CFDs) raise money through special taxes that must be approved by 2/3rds of the voters within the district. A CFD is formed to finance major improvements and services within the district which might include schools, roads, libraries, police and fire protection services, or ambulance services. The taxes are secured by a continuing lien and are levied annually against property within the district.

The Office of Public Finance intends to sell the GO Bonds through a competitive sale process, but in the case of significant change in market conditions, reserves the option to seek a negotiated sale with underwriter(s) selected competitively.

Debt Limit

Section 9.106 of the City Charter limits the amount of GO bonds the City can have outstanding at any given time to 3 percent of the total assessed value of property in San Francisco. The City calculates its debt limit on the basis of total assessed valuation net of non-reimbursable and homeowner exemptions. On this basis, the City's gross general obligation debt limit for FY 2018-19 is approximately \$7.78 billion, based on a net assessed valuation of approximately \$259.3 billion.

As of March 1, 2019, the City had outstanding approximately \$2.53 billion in aggregate principal amount of GO bonds, which equals approximately 0.97 percent of the net assessed valuation for FY 2018-19. If the Board of Supervisors approves the issuance of the Series 2019B GO Bonds, the debt ratio would increase by 0.02 percentage points to 0.99 percent — within the 3 percent legal limit. If all of the City's authorized and unissued bonds were issued, the total debt burden would be 1.40 percent of the net assessed value of property in the City.

Property Taxes

For Series 2019B, repayment of the annual debt service will be recovered through increases in the annual Property Tax rate, which, according to the Controller's Office, would be \$0.00132 per \$100 of assessed value or \$1.32 per \$100,000 of assessed value over the anticipated 20-year term of the bonds. The owner of a residence with an assessed value of \$600,000, assuming a homeowner's exemption of \$7,000, would pay average additional property taxes to the City of approximately \$7.81 per year if the anticipated \$47,010,000 Bonds are sold.

Capital Plan

Under financial constraints adopted by the City's Capital Planning Committee, debt service on approved and issued GO bonds may not increase property owners' long-term property tax rates above FY 2005-06 levels. The FY 2005-06 property tax rate for the GO bond fund was \$0.1201 per \$100 of assessed value. If the Board of Supervisors approves the issuance of the Series 2019B GO Bonds, the property tax rate for GO bonds for FY 2018-19 would remain below the FY 2005-06 rate and within the Capital Planning Committee's approved financial constraint.

POLICY CONSIDERATION

According to Mr. Vishal Trivedi, Financial Analyst in the Office of Public Finance, there has been a delay in hearing this legislation due to a court case challenging the validity of the bond measure. On June 19, 2019, the Superior Court of the State of California ruled in favor of the City and dismissed the suit. However, because the 60-day period to file an appeal does not end until August 19, 2019, the Office of Public Finance does not plan to sell the bonds until after that date.

RECOMMENDATIONS

- 1. Request the Office of Public Finance to submit a memorandum to the Board of Supervisors on the status of the potential appeal after the close of the 60-day period to appeal.
- 2. Approve the proposed ordinance and resolutions.

PUBLIC UTILITIES COMMISSION

City and County of San Francisco

RESOLUTION NO.	17-0199

WHEREAS, The California Air Resources Board (CARB) adopted the Low Carbon Fuel Standard Program (LCFS Program) in 2009 to reduce by 10 percent the 2010 baseline level for carbon intensity (CI) of transportation fuels used in California (State Target). The applicable regulations are found in the California Code of Regulations, Title 17, Sections 95480, et seq. (CARB Regulations); and

WHEREAS, The San Francisco Public Utilities Commission (SFPUC) provides the San Francisco Municipal Transportation Agency (SFMTA) with greenhouse gas-free electric energy to operate its electric public transit fleet, pursuant to SFPUC rate schedule M-2; and

WHEREAS, Using the electricity provided by the SFPUC, the SFMTA operates its electric public transit fleet (electric trolley buses, cable cars and light-rail vehicles) with a CI below the State Target, and therefore accumulates credits, which can be sold via the CARB's market place to energy producers, such as refiners and fuel importers, with a CI above the State Target in order to balance State deficits. Each credit represents one metric ton of CO2 equivalent (greenhouse gas) emissions avoided; and

WHEREAS, The SFMTA registered for the LCFS Program in September 2016 and began accumulating credits on a quarterly basis, starting in the fourth quarter of calendar year 2016; and

WHEREAS, In coordination with SFMTA, the SFPUC will sell credits in order to fund projects that will (1) advance the SFMTA's sustainability and climate action initiatives and goals, as described in the SFMTA Strategic Plan for the transportation sector Climate Action Strategy, the Departmental Climate Action Plan, the SFMTA Clean Air Plan, and the agency's APTA Sustainability Commitment (Green Projects); and (2) support the SFPUC's clean energy operations; and

WHEREAS, The estimated revenue to the City and County of San Francisco from the LCFS program is \$1 million to \$2 million per year and; and

WHEREAS, To realize this value the SFPUC and SFMTA propose to enter into a Memorandum of Understanding (MOU) where net revenues from the LCFS program will be shared 50/50 between the SFPUC and SFMTA; and

WHEREAS, The proposed MOU assigns responsibilities for participation in the LCFS program between the SFPUC and SFMTA, and establishes safeguards to limit program administration costs; and

WHEREAS, This action would not result in a physical change in the environment and therefore does not constitute a "project" under California Environmental Quality Act Guidelines Section 15378; now, therefore, be it

RESOLVED, That this Commission hereby authorizes the Power Enterprise of the San Francisco Public Utilities Commission to sell credits through the California Air Resources Board (CARB) Low Carbon Fuel Standard Program (LCFS Program) to support the SFPUC's clean energy operations, and delegates to the General Manager authority to enter agreements necessary to effectuate the sales and to seek Board of Supervisors approval, if needed, of any sales under this Program in excess of \$1 million per Charter Section 9.118(a); and be it

FURTHER RESOLVED, That this Commission hereby authorizes the General Manager of the San Francisco Public Utilities Commission to execute a Memorandum of Understanding with the San Francisco Municipal Transportation Agency r regarding the City's participation in the Low Carbon Fuel Standard Program; and be it

FURTHER RESOLVED, That this Commission authorizes the MOU for a term not-to-exceed five (5) years, with an option to renew for an additional five (5) years, subject to approval of this Commission.

I hereby certify that the foregoing resolution was adopted by the Public Utilities Commission at its meeting of September 12, 2017.

Secretary, Public Utilities Commission

MEMO

DATE:

November 22, 2017

TO:

Theresa Mueller

Deputy City Attorney

FROM:

Rosie Scott
Record Retention Coordinator

SEPUC Power

RE:

MOU between SFPUC and SFMTA

Low Carbon Fuel Standard Program (LCSF)

Please find attached signed version of the Memorandum of Understanding between the SFPUC and SFMTA and the resolution approved by the SF Public Utilities Commission. The California Air Resources Board (CARB) adopted the Low Carbon Fuel Standard Program in 2009 to reduce by 10 percent the 2010 baseline level for carbon intensity of transportation fuels use in California.

Enclosure: Attachment

MEMORANDUM OF UNDERSTANDING

between

THE SAN FRANCISCO MUNICIPAL TRANSPORTATION AGENCY

and

THE SAN FRANCISCO PUBLIC UTILITIES COMMISSION

THIS MEMORANDUM OF UNDERSTANDING (this MOU) is made this _____day of September, 2017, in the City and County of San Francisco, State of California, by and between the San Francisco Municipal Transportation Agency (SFMTA) and the San Francisco Public Utilities Commission (SFPUC) (collectively, "Parties").

RECITALS

- A. The California Air Resources Board (CARB) adopted the Low Carbon Fuel Standard Program (LCFS Program) in 2009 to reduce by 10 percent the 2010 baseline level for carbon intensity (CI) of transportation fuels used in California (State Target). The applicable regulations are found in the California Code of Regulations, Title 17, Sections 95480, et seq. (CARB Regulations). The LCFS Program is expected to continue through at least 2022.
- B. AB 162 (Statutes of 2009) and Senate Bill 1305 (Statutes of 1997) require retail electricity suppliers to disclose information to California consumers about the energy resources used to generate the electricity they sell. Based on SFPUC's 2016 Power Content Label, SFPUC's resource portfolio consisted of 100% greenhouse-gas (GHG) free resources. San Francisco has reported being 100% GHG-free since 2011 when it began its Power Content Label reporting to consumers.
- C. The SFPUC provides the SFMTA with greenhouse gas-free electric energy to operate its electric public transit fleet, pursuant to SFPUC rate schedule M-2.
- D. Using the electricity provided by the SFPUC, the SFMTA operates its electric public transit fleet (electric trolley buses, cable cars and light-rail vehicles) with a CI below the State Target, and therefore accumulates credits, which can be sold via the CARB's market place to energy producers, such as refiners and fuel importers, with a CI above the State Target in order to balance State deficits. Each credit represents one metric ton of CO₂ equivalent (greenhouse gas) emissions avoided. The market-based value of credits has continued to increase since the Program began.
- E. The SFMTA registered for the LCFS Program in September 2016 and began accumulating credits on a quarterly basis, starting in the fourth quarter of calendar year 2016. The estimated value of the credits accumulated by the SFMTA to date is roughly \$680,000. Estimated revenues could be between \$1 and \$2 million per year at current prices.
- F. The SFMTA and the SFPUC desire to sell credits in order to fund projects that will (1) advance the SFMTA's sustainability and climate action initiatives and goals, as described in the SFMTA Strategic Plan for the transportation sector Climate Action Strategy, the Departmental Climate Action Plan, the SFMTA Clean Air Plan, and the agency's APTA Sustainability Commitment (Green Projects); and (2) support the SFPUC's clean energy operations.

NOW, THEREFORE, the SFMTA and the SFPUC agree as follows:

I. Responsibilities of SFMTA and SFPUC

A. Creation of Workplan: The SFMTA and the SFPUC will annually create a Workplan identifying how each agency will achieve completion of the following tasks while minimizing costs and avoiding duplication of effort.

B. SFMTA

- 1. Calculation of Credits. The SFMTA, as Account Administrator, will submit the necessary quarterly and annual reports to CARB to calculate the amount of credits available for sale under the LCFS Program in accordance with CARB regulations. The calculation will take into consideration disaggregation of vehicle electricity usage from total meter readings and other technical issues.
- 2. Regulatory changes. The SFMTA will be responsible for keeping the Parties up to date on the CARB regulations, including notifying the Parties of the changes in a timely manner.

C. SFPUC

- 1. Billing Support. The SFPUC will provide meter and billing data and technical support in a timely manner to allow the SFMTA to submit required information regarding energy usage to CARB to receive credits; provide supplemental documentation as needed; and meet CARB data retention requirements associated with the creation of credits.
- Credit Sales. The SFPUC will monitor the market and manage the sale of credits under the LCFS Program in compliance with the CARB regulations and the strategies jointly developed by the SFPUC and SFMTA in Section D.1.
- 3. Reporting. The SFPUC will provide the administrative support needed for the sale of credits, track and record sales revenues, and provide necessary data to SFMTA to allow it to prepare the quarterly and annual reports, as required under the CARB Regulations. The SFPUC will also report to the SFMTA after each sale has closed, with a description of the amount of the sale, the number of credits sold, and, if available, the identity of the purchaser of the credits.

D. Joint Tasks

- 1. Credit Sales. The SFMTA and the SFPUC will coordinate in developing a mutually acceptable strategy, method, and timing for the sale of credits.
- 2. Legislative/Regulatory Changes. The SFMTA and the SFPUC will coordinate in developing mutually acceptable strategies for the introduction or negotiation of any legislative or regulatory changes to the LCFS Program.

- 3. Public Meetings. The SFMTA and the SFPUC will collaborate in developing mutually acceptable presentations for their respective boards, for other City boards and commissions, and for other public meetings.
- 4. Outreach. The SFMTA and the SFPUC will conduct outreach, public relations, and media activities for the LCFS Program.
- 5. Coordination. The SFMTA and the SFPUC will consult and agree on any significant expenses associated with the Program. This includes, but is not limited to, retention of outside consulting or brokerage services, and extensive participation in a regulatory or legislative proceeding
- 6. SFPUC-specific LCFS Pathway. The SFMTA and SFPUC will work together to allow the SFPUC to file a SFPUC-specific LCFS Pathway application to CARB reflecting the SFPUC's GHG emission profile...
- 7. Cost Tracking. The SFPUC and the SFMTA will each track administrative and other program costs on an hourly or actual cost basis (including overhead) using standard City Work Orders and tracking of time through City index codes.

II. Allocation of Net Revenues

- A. Net Revenues. "Net Revenues" means proceeds from the sale of credits less Administrative Costs.
- B. Administrative Costs. Administrative and other program costs are to be paid from the proceeds from the sale of credits. The Parties will seek to keep costs within any estimates prepared as part of the Workplan, but in no event may the Administrative Costs for each Party exceed five percent of the total revenues from the sale of credits.
- C. Allocation Between Parties. Net Revenues will be allocated 50% to SFMTA and 50% to the SFPUC.
- D. SFMTA Use of Net Revenues. The SFMTA will manage the disbursement of the Net Revenues allocated to the SFMTA to Green Projects.
- E. SFPUC Use of Net Revenues. The SFPUC will use its share of Net Revenues for its clean energy operations.

III. Term and Termination.

- A. Base Term. The term of the MOU will commence on the last date of signature below and end five years thereafter.
- B. Option. There will be an option to renew the MOU for an additional five years, to be exercised through an amendment to this MOU.
- C. Termination. This MOU may be terminated for any reason by either party upon 90-days' notice from the party terminating the MOU to the non-terminating party. A final Allocation of Net Revenues will be performed and settled not later than 10

days after receipt of the proceeds from the sale of credits following notice of termination.

IV. Disputes

Any disputes regarding the provisions of this MOU that cannot be resolved by the Chief Financial Officers of each Agency shall be elevated to the Director of Transportation of the SFMTA and the General Manager of the SFPUC.

V. Miscellaneous

- A. Modification. This MOU may not be modified, nor may compliance with any of its terms be waived, except by written instrument executed and approved in the same manner as this MOU, except that the SFPUC General Manager may approve non-material amendments consistent with the intent of this MOU.
- B. Notices. Unless otherwise specifically provided herein, all notices, consents, directions, approvals, instructions, requests and other communications hereunder shall be in writing, shall be addressed to the person and address set forth below and shall be deposited in the U.S. mail or emailed.

To SFPUC:

San Francisco Public Utilities Commission Attn: Barbara Hale, Assistant General Manager, Power 525 Golden Gate Avenue, 13th Floor San Francisco, CA 94102 Email: bhale@sfwater.org

To SFMTA:

San Francisco Municipal Transportation Agency Attn: Sonali Bose, Chief Financial Officer One South Van Ness Avenue, 3rd floor San Francisco, CA 94103 Email: sonali.bose@sfmta.com

IN WITNESS WHEREOF, the Parties have caused this MOU to be duly executed as of the date first specified herein.

SAN FRANCISCO PUBLIC UTILITIES COMMISSION

SAN FRANCISCO MUNICIPAL TRANSPORTATION AGENCY

HÁRĽAN L. KELLY, JR.

GENERAL MANAGER

On: 10-6-17

DIRECTOR OF TRANSPORTATION

LEAP MASTER AGREEMENT FOR PURCHASING AND SELLING LOW CARBON FUEL STANDARD CREDITS Version 1.1

Effective	as	of	

COVER SHEET

This LEAP Master Agreement for Purchasing and Selling Low Carbon Fuel Standard Credits Version 1.1 ("LEAP LCFS Agreement") together with, and as amended by, this Cover Sheet and any annexes, schedules and written supplements hereto, and all Transactions (including any Confirmations) shall be referred to collectively as this "Agreement".

The parties to this Agreement are as follows: ("Party A") ("Party B") U.S. Federal Tax ID No.: U.S. Federal Tax ID No.: Notices: Notices: Mail address: Mail address: Attn:_____ Attn: Phone: Fax: Phone:_____Fax:____ Wire Transfer or ACH Numbers (if applicable): Wire Transfer or ACH Numbers (if applicable): ABA: ACCT: ACCT: Notices for Default: Notices for Default: Attn: Attn: Phone: Fax: Phone:_____Fax:____ Other Details: Other Details: The Parties hereby agree that the General Terms and Conditions of this LEAP Master Agreement for Purchasing and Selling Low Carbon Fuel Standard Credits set forth below are incorporated herein subject to the following elections and modifications: Section 8 - Events of Default THE FOLLOWING EVENTS OF DEFAULT ARE APPLICABLE UNDER THIS AGREEMENT (CHECK ALL PROVISIONS THAT ARE APPLICABLE): 8.1(d) Default Under Other Trading Agreements: [] Applicable. If not checked, inapplicable. "Aggregate Delinquency Amount" means for purposes of Section 8.1(d) (if applicable): With respect to Party A, Party A's Credit Support Provider and each applicable Specified Affiliate of Party With respect to Party B, Party B's Credit Support Provider and each applicable Specified Affiliate of Party "Specified Affiliate" means for purposes of Section 8.1(d) (if applicable): With respect to Party A: With respect to Party B: 8.1(e) Cross Default: [] Applicable. If not checked, inapplicable. Cross Acceleration. If Section 8.1(e) is applicable, the words ", or becoming capable at such time of being declared," in Section 8.1(e) shall be deleted. If not checked, these words shall not be deleted. "Threshold Amount" means for purposes of Section 8.1(e) (if applicable): With respect to Party A:

With respect to Party B:

Section 9 – Termination and Liquidation 9.3 Closeout Setoff:					
For purposes of Section 9.3, the Parties agree that (select one):					
[] Option A (Bilateral Setoff) is applicable. (Applicable if no other election is made.)					
Option B (Triangular Setoff) is applicable.Option C (Rectangular Setoff) is applicable.					
[] Option D (No Setoff) is applicable.					
Section 10 – Force Majeure					
10.5 Force Majeure Termination Payment:					
[] Applicable. If not checked, inapplicable. If applicable, the following provi	sions will apply: on Payment will be payable in respect of Term Transactions that are				
	in two (2) New York Banking Days of termination. The Termination				
Payment and the Party that owes such Termination Payment will be	e calculated and determined pursuant to the methodology set forth in				
	Party will be deemed the "Performing Party" for purposes of applying nly the Transactions being terminated pursuant to Section 10.5 will be				
taken into account in determining the amount of such payment and	all other Transactions will remain outstanding and unaffected by such				
termination.					
one-half of the difference between the Termination Payment of the Payment of the Party with the lower Termination Payment ("Y"). If	calculate a Termination Payment, and the amount payable will equal Party with the higher Termination Payment (" \underline{X} ") and the Termination the amount payable is a positive number, Y will pay it to X; if it is a				
negative number, X will pay the absolute value of that amount to Y.					
Section 19 - New or Changed Laws or Regulations					
19.1 Material Adverse Effect:					
[] Applicable. If not checked, inapplicable.					
19.2 <u>Disrupted Transactions</u> :					
[] Applicable. If not checked, inapplicable.					
Section 18 – Written Confirmations					
Confirming Party:					
[] Party A [] Party B					
Other Modifications to the LEAP LCFS Agreement					
Specify, if any: Schedule to the LEAP LCFS Agreement					
IN WITNESS WHEREOF, the Parties have caused this Agreement to be	duly executed as of the date first above written.				
and					
("Party A")	("Party B")				
Ву:	Ву:				
Name:	Name:				
Title:	Title:				
Date:	Date:				
DISCLAIMER					

THE OBJECTIVE OF THIS AGREEMENT IS TO FACILITATE THE ORDERLY AND PROMOTE THE EFFICIENT TRADING OF LCFS CREDITS. USE OF THIS AGREEMENT OR ANY PROVISION THEREOF IS COMPLETELY VOLUNTARY. USE OF THIS AGREEMENT IS NOT RESTRICTED AND LEAP HEREBY AUTHORIZES ANYONE TO USE THIS AGREEMENT OR ANY PROVISION THEREOF.

NONE OF LEAP, ANY LEAP MEMBER COMPANY OR ANY OF THEIR AGENTS, REPRESENTATIVES OR ATTORNEYS SHALL BE RESPONSIBLE FOR THE USE OF THIS AGREEMENT, OR ANY DAMAGES OR LOSSES RESULTING THEREFROM. BY PROVIDING THIS AGREEMENT, LEAP DOES NOT OFFER LEGAL ADVICE AND ALL USERS ARE URGED TO CONSULT THEIR OWN LEGAL COUNSEL AND TAX ADVISORS TO ENSURE THAT THIS AGREEMENT WILL ACHIEVE THEIR COMMERCIAL OBJECTIVES AND THEIR LEGAL INTERESTS ARE ADEQUATELY PROTECTED.

GENERAL TERMS AND CONDITIONS

In consideration of the mutual undertakings in this Agreement and for other good and valuable consideration, the receipt and sufficiency of which the Parties acknowledge, the Parties agree as follows:

1. DEFINITIONS AND INTERPRETATION

1.1 Definitions

- (a) "Accepted" or "Acceptance" has the meaning specified in Section 2.2.2.
- (b) "Affected Party" has the meaning specified in Section 19.1.
- (c) "Affected Transaction" has the meaning specified in Section 19.1.
- (d) "Affiliate" means, in relation to any person, an entity controlled, directly or indirectly, by the person, any entity that controls, directly or indirectly, the person or any entity directly or indirectly under common control with the person. For this purpose, "control" of any entity or person means ownership of a majority of the voting power of the entity or person.
- (e) "Aggregate Delinquency Amount" means, with respect to a Party, the amount specified as the Aggregate Delinquency Amount for such Party in respect of Section 8.1(d) in the Cover Sheet.
- (f) "Agreement" has the meaning set out in the preamble to the Cover Sheet.
- (g) "Applicable Law" means any federal, national, state or local law, statute, regulation, code, ordinance, license, permit, compliance requirement, decision, order, writ, injunction, directive, judgment, policy, decree, including any judicial or administrative interpretations thereof, or any agreement, concession or arrangement with any Governmental Authority, applicable to either Party or either Party's performance under any Transaction, and any amendments or modifications to the foregoing.
- (h) <u>"Alternative Settlement Amount"</u> has the meaning specified in Section 19.1.
- (i) "ARB" means the California Air Resources Board or successor agency.
- "Bankrupt" means, with respect to a Party, that such Party: (i) is dissolved (other than pursuant to a consolidation, amalgamation (i) or merger); (ii) becomes insolvent or is unable to pay its debts or fails or admits in writing its inability generally to pay its debts as they become due; (iii) makes a general assignment, arrangement or composition with or for the benefit of its creditors; (iv)(A) institutes, or has instituted against it by a regulator, supervisor or any similar official with primary insolvency, rehabilitative or regulatory jurisdiction over it in the jurisdiction of its incorporation or organization or the jurisdiction of its head or home office, a proceeding seeking a judgment of insolvency or bankruptcy or any other relief under any bankruptcy or insolvency law or other similar law affecting creditors' rights, or a petition is presented for its winding-up or liquidation by it or such regulator, supervisor or similar official, or (B) has instituted against it a proceeding seeking a judgment of insolvency or bankruptcy or any other relief under any bankruptcy or insolvency law or other similar law affecting creditors' rights, or a petition is presented for its winding-up or liquidation, and such proceeding or petition is instituted or presented by a person or entity not described in clause (A) above; (v) has a resolution passed for its winding-up, official management or liquidation (other than pursuant to a consolidation, amalgamation or merger); (vi) seeks or becomes subject to the appointment of an administrator, provisional liquidator, conservator, receiver, trustee, custodian or other similar official for it or for all or substantially all its assets; (vii) has a secured party take possession of all or substantially all its assets or has a distress, execution, attachment, sequestration or other legal process levied, enforced or sued on or against all or substantially all its assets; (viii) causes or is subject to any event with respect to it which, under the Applicable Law, has an analogous effect to any of the events specified in clauses (i) to (vii) (inclusive); or (ix) takes any action in furtherance of, or indicating its consent to, approval of, or acquiescence in, any of the foregoing acts.
- (k) "Bankruptcy Code" has the meaning specified in Section 9.7.
- (I) "Buyer" means the Party obligated to purchase or acquire LCFS Credits under a Transaction.
- (m) "Confirmation" means (i) any electronic confirmation setting forth the trade details of a Transaction (and any special conditions) between the Parties and matched by the Parties on an electronic confirmation matching system and (ii) absent the ability to confirm a Transaction through an electronic confirmation matching system, any other written or electronic confirmation between the Parties that contains the relevant trade details (and any special conditions) of the Transaction, in either case based on those set forth in the sample form attached as Exhibit 1.
- (n) "Contract Price" means the price (expressed in U.S. Dollars) of an LCFS Credit as specified in a Confirmation.
- (o) "Contract Value" means the number of the LCFS Credits remaining to be delivered or purchased under a Transaction multiplied by the Contract Price.
- (p) "Credit Support Provider" means a Party's guarantor or other person providing Performance Assurance for such Party.

- "Credit Transfer Form" means the Credit Transfer Form that is incorporated by reference into the LCFS Regulations in 17 CCR (q) §95487(c)(1)(B), properly completed and executed by Seller in accordance with the LCFS Regulations.
- "Defaulting Party" has the meaning specified in Section 9.1. (r)
- "Deficient LCFS Credit" has the meaning specified in Section 7.1. (s)
- "Designated Event" means, with respect to a Party for purposes of Section 8.1(f); (i) the consolidation or amalgamation of a Party (t) with, the merger of a Party with or into, or the transfer of all or substantially all of a Party's assets to, another entity; (ii) the reorganization, reincorporation or reconstitution of a Party into or as another entity; (iii) the acquisition by any person directly or indirectly of the majority of the beneficial ownership of the Party such that such person may exercise control of the Party; or (iv) a substantial change in the capital structure of a Party by means of the issuance or guaranty of debt.
- (u) "Disrupted Transaction" has the meaning specified in Section 19.2.
- "Early Termination Date" has the meaning specified in Section 9.1. (v)
- "Eastern Prevailing Time" means the time prevailing on the East Coast of the U.S., taking into account daylight savings time if it (w) is in effect.
- "Event of Default" has the meaning specified in Section 8.1. (x)
- "Executive Officer" has the meaning specified in the LCFS Regulations. (y)
- (z) "Expert" means a person reasonably agreed upon by the Parties who shall be qualified by education, experience or training to resolve the applicable issue and who shall not be a current or former employee of either Party or otherwise have a stake in the outcome of the dispute.
- "Expert's Decision" has the meaning specified in Section 19.1. (aa)
- "Force Majeure" has the meaning specified in Section 10.1. (bb)
- "Force Majeure Termination Payment" has the meaning specified in Section 10.5. (cc)
- "Governmental Authority" means any U.S. federal, state, regional, local or municipal governmental body, agency, (dd) instrumentality, authority or entity established or controlled by a government or subdivision thereof, including any legislative, administrative or judicial body, or any person acting on behalf thereof.
- "Independent Dealer" means a leading dealer in the relevant physical trading markets for the Affected Transactions that is not a (ee) Party or an Affiliate of a Party.
- "Inappropriate Payments" has the meaning specified in Section 20.4. (ff)
- "Initiate" means the submission of a sell transaction in the LRT-CBTS_by Seller provided, however, that a Seller shall not be (gg) deemed to have submitted any LCFS Credits where Seller cancels such sell transaction in the LRT-CBTS before Buyer accepts it in the LRT-CBTS.
- "Invoice" has the meaning set forth in Section 4.1. (hh)
- "LCFS Account" means the account of a Party showing the LCFS Credits and LCFS Deficits generated by the Party or (ii)transferred, purchased or acquired by the Party, as established with ARB or another governmental authority pursuant to the LCFS Regulations.
- "LCFS Regulations" means the regulations, orders, decrees and standards issued by a governmental authority implementing or (jj) otherwise applicable to the California Low Carbon Fuel Standard as set forth in 17 CCR § 95480 et seg. and each successor regulation, as may be subsequently amended, modified, restated from time to time.
- (kk) "LCFS Credit" means a Credit as defined in the LCFS Regulations.
- "LCFS Deficit" means a deficit as defined in the LCFS Regulations. (II)
- "LRT-CBTS" means "LCFS Reporting Tool and Credit Bank and Transfer System". (mm)
- "LRT Transfer Notification" has the meaning specified in Section 2.2.1. (nn)

- (oo) "LEAP" has the meaning specified in the disclaimer paragraph on the last page of the Cover Sheet.
- (pp) "LEAP LCFS Agreement" has the meaning specified in the Cover Sheet.
- (qq) "Market Value" means the amount of the LCFS Credits remaining to be Initiated under a Transaction multiplied by the market price for an equivalent transaction for Qualified Replacement LCFS Credits as determined by the Performing Party (or determining party, as the case may be) in a commercially reasonable manner. To ascertain the Market Value, the Performing Party may consider, among other valuations, quotations from leading dealers in swap contracts or physical trading markets, similar sales or purchases and any other bona fide third-party offers, all adjusted for the length of the term, relevant Payment Due Dates, Transfer Dates, and Transaction Volume. A Party shall not be required to enter into a replacement transaction in order to determine the Market Value of a Transaction. For the avoidance of doubt, any option pursuant to which one Party has the right to extend the term of a Transaction shall be considered in determining Contract Value and Market Values.
- (rr) "New York Banking Day" means a day (other than a Saturday or Sunday or banking holiday) on which commercial banks are authorized to open for business in the State of New York.
- (ss) "Original Index" has the meaning specified in Section 19.2.
- (tt) "Other Amounts" has the meaning specified in Section 9.3.
- (uu) "Other Trading Agreement" has the meaning specified in Section 8.1(d).
- (vv) "Party" means Party A or Party B, individually, and "Parties" means Party A and Party B, collectively.
- (ww) "Party A" is the party designated as such in the introduction on page 1 of the Cover Sheet
- (xx) "Party B" is the party designated as such in the introduction on page 1 of the Cover Sheet.
- (yy) "Payment Due Date" means the payment due date specified in the Confirmation (or otherwise agreed in writing by the Parties), provided that if the Payment Due Date is not so specified or agreed, then it shall be five (5) New York Banking Days after the later of (A) the Transfer Date or (B) the payer's receipt of the payee's Invoice.
- (zz) "Pending Credits" has the meaning specified in Section 7.3.
- (aaa) "Performance Assurance" has the meaning specified in Section 6.1.
- (bbb) "Performing Party" has the meaning specified in Section 9.1.
- (ccc) "Physical Pathway Code" has the meaning specified in the LCFS Regulations in 17 CCR §95481(a)(40).
- (ddd) "Posting Party" has the meaning specified in Section 6.1.
- (eee) "Prepayment" has the meaning specified in Section 6.2.
- (fff) "Prepayment Amount" has the meaning specified in Section 6.2.
- (ggg) "Prepayment Due Date" has the meaning specified in Section 6.2.
- (hhh) "Qualified Institution" has the meaning specified in Section 6.1(b).
- (iii) "Qualified Replacement LCFS Credit" means a valid LCFS Credit meeting the specifications set forth in the relevant Confirmation.
- (jjj) "Quantity" means, with respect to a Transfer Date, the number of LCFS Credits to be purchased and sold pursuant to a Transaction as specified in the Confirmation.
- (kkk) "Reference Price" means a price that is determined by reference to a specified pricing source.
- (III) "Renegotiation Notice" has the meaning specified in Section 19.1.
- (mmm) "Renegotiation Option" has the meaning specified in Section 19.1.
- (nnn) "Secured Party" has the meaning specified in Section 6.1.
- (000) "Seller" means the Party obligated to sell or Transfer LCFS Credits under a Transaction.
- (ppp) "Specified Affiliate" means, with respect to a Party, the entity or entities specified as the Specified Affiliate(s) with respect to such Party in respect of Section 8.1(d) in the Cover Sheet.

- (qqq) "Termination Date" has the meaning specified in Section 19.1.
- (rrr) "Termination Payment" has the meaning specified in Section 9.2.
- (sss) "Termination Value" means the aggregate value of all Affected Transactions determined by valuing each Affected Transaction at its Market Value as reasonably determined by the determining party as of the Termination Date and then determining the amount by which such then prevailing Market Value differs from the Contract Value (it being understood that (i) in the event the prevailing Market Value of an Affected Transaction exceeds the Contract Value, the difference in value shall be due from Seller to Buyer, and (ii) in the event that the prevailing Market Value of an Affected Transaction is less than the Contract Value, the difference in value shall be due from Buyer to Seller) and adding to such value, without duplication, any amounts then payable under the Affected Transaction (including amounts due in respect of LCFS Credits Initiated and Accepted hereunder). The terms "Market Value" and "Contract Value" as used in this definition shall have the meanings specified in Section 1, provided that each reference to "Performing Party" shall be replaced by "determining party".
- (ttt) "Term Transaction" means a Transaction contemplating multiple Transfers with a delivery period the duration of which is one (1) calendar month or more.
- (uuu) "Threshold Amount" has the meaning specified in the Cover Sheet in respect of Section 8.1(e).
- (vvv) "Trade Date" means the date a Transaction is entered into between the Parties.
- (www) "Transaction" has the meaning specified in Section 1.3.
- (xxx) "Transfer" or "Transferred" has the meaning specified in Section 2.3.
- (yyy) "Transfer Date" has the meaning specified in Section 2.3 (b)
- (zzz) "Transfer Obligations" has the meaning specified in Section 2.2.1.
- (aaaa) "<u>Transfer Period</u>" means, for a Transaction, the date range as specified in the Confirmation during which Seller must Initiate the Quantity of LCFS Credits specified in the Confirmation, all in accordance with the Seller's obligations under the LCFS Regulations, and any subsequent guidance from ARB.
- (bbbb) "<u>U.S.</u>" means United States of America, and every reference to money, price, or Contract Price pertains to U.S. Dollars.

1.2 <u>Interpretation</u>

Unless otherwise specified, all section references in this Agreement are to the Sections of this Agreement. All headings in this Agreement are intended solely for convenience of reference and shall not affect the meaning or interpretation of this Agreement. Unless expressly provided otherwise, the word "including" as used herein does not limit the preceding words or terms and shall be read to be followed by the words "without limitation" or words having similar import and the words "other" and "otherwise" shall not be construed as being limited by the context in which they appear or the words that precede them. The word "or" is not exclusive. Unless otherwise expressly stated, the words "hereof", "herein" and "hereunder" and words of similar import refer to this Agreement as a whole and not to any particular provision of this Agreement. The words "will" and "shall" are expressions of command, not merely expressions of future intent or expectation. Unless expressly provided otherwise, references to "consent" mean the prior written consent of the Party at issue. Unless provided otherwise, when a Party's response is required hereunder within a specific time period following receipt of notice or documentation, as applicable, the day of receipt thereof by such Party shall be considered day zero. The Parties acknowledge that they and their counsel have reviewed and revised this Agreement and that no presumption of contract interpretation or construction shall apply to the advantage or disadvantage of the drafter of this Agreement. Any specific references to laws, statutes, or regulations will include any amendments, replacements, or modifications thereto.

1.3 Scope; Single Agreement

This Agreement is intended to apply exclusively to transactions for the purchase and sale of LCFS Credits between Seller and Buyer, the details of which are set forth in a Confirmation (a "<u>Transaction</u>"). All Transactions are entered into in reliance on the fact that this Agreement and all Transactions hereunder form a single agreement between the Parties.

1.4 Inconsistency

In the event of any inconsistency between the provisions of any Confirmation and this Agreement, such Confirmation will prevail for the purposes of the relevant Transaction; provided however, any changes which conflict with any provisions of this Agreement regarding Section 6: Credit, Section 8: Events of Default, and Section 9: Termination and Liquidation must be agreed in writing by both Parties. For the avoidance of doubt, any repetition in a Confirmation of any section or any part of such section of this LEAP LCFS Agreement shall be for emphasis only and shall not by reason of such repetition exclude any other part of such section or any other section or any part thereof of this LEAP LCFS Agreement unless such intention to override the LEAP LCFS Agreement is explicitly expressed in the Confirmation.

2. GENERAL PURCHASE AND SALE OBLIGATIONS; TITLE TRANSFER

2.1 Purchase

Pursuant to each Transaction, and subject to the terms and conditions of the Confirmation governing such Transaction and this LEAP LCFS Agreement, Seller agrees to sell and Initiate to Buyer the Quantity of LCFS Credits at the Contract Price during the Transfer Period, all as specified in the Confirmation, and Buyer agrees to purchase and Accept the LCFS Credits from Seller, subject to its rights under Section 2.4, within the specified number of days required by the LCFS Regulations.

2.2.1 <u>Initiation</u>

Each Party shall take all actions required by the LCFS Regulations to effect a transfer of the LCFS Credits from Seller to Buyer during the Transfer Period (the "Transfer Obligations"). The LCFS Credits shall be deemed initiated ("Initiated") by Seller to Buyer upon:

- (a) Buyer's receipt from Seller of a Credit Transfer Form or a notification of an electronic transfer of LCFS Credits via the LRT-CBTS (each such notification, a "LRT Transfer Notification"); and
- (b) Seller's satisfaction of all other Transfer Obligations applicable to Seller, if any.

Seller shall Initiate LCFS Credits on or before the end of the Transfer Period as indicated on the Confirmation.

2.2.2 Acceptance

LCFS Credits that are Initiated by Seller shall be deemed accepted ("Accepted") by Buyer upon:

- (a) either Buyer's submission of the Credit Transfer Form to the Executive Officer in accordance with the LCFS Regulations or Buyer's acceptance, via the LRT-CBTS, of a LRT Transfer Notification; and
- (b) Buyer's satisfaction of all other Transfer Obligations applicable to Buyer, if any; and Buyer's submission of the Credit Transfer Form or Buyer's acceptance of a LRT Transfer Notification, together with the satisfaction of all other Buyer Transfer Obligations, if any, shall constitute the acceptance (the "Acceptance") by Buyer of the LCFS Credits.

2.3 Title Transfer

Title to the LCFS Credits shall be deemed to transfer from Seller to Buyer after

- (a) Initiation and Acceptance of the LCFS Credits; and
- (b) upon the recordation of the addition of the LCFS Credits to the LCFS Account balance of Buyer and the subtraction of the LCFS Credits from the LCFS Account balance of Seller by the Executive Officer (the title transfer of LCFS Credits as set forth above hereafter referred to as the "Transfer" of the LCFS Credits and the date on which the Transfer occurs is the "Transfer Date").

2.4 Buyer Right to Reject

Buyer shall have the right, at its reasonable discretion, to reject any LCFS Credits Initiated by Seller upon notice to Seller within the specified number of days required by the LCFS Regulations. For the avoidance of doubt, and without limitation, Buyer shall be conclusively deemed to have reasonably exercised its discretion to reject where:

- (a) Buyer reasonably believes that the Credit Transfer Form or the information in a LRT Transfer Notification does not reflect the terms of the Transaction:
- (b) the LCFS Credits are invalid under the LCFS Regulations or are subject to a proceeding by a governmental authority or are otherwise encumbered;
- (c) there is a reasonable prospect that the LCFS Credits will be invalid under the LCFS Regulations; or
- (d) Buyer does not have information sufficient to verify: (i) that any of the LCFS Credits are valid, and (ii) that there is no reasonable prospect of such LCFS Credits becoming invalid under the LCFS Regulations.

For purposes of making its assessment it shall be reasonable for Buyer to disregard the benefit of any warranties given to it under this Agreement.

Without prejudice to the application of Section 10, it is not a reasonable exercise of discretion for Buyer to reject LCFS Credits solely on the basis of market conditions and/or the market price of, LCFS Credits.

2.5 Recordation by the Executive Officer

Upon notification from the Executive Officer that any transfer of all or a portion of the LCFS Credits will not be recorded, the Parties shall promptly confer and shall cooperate in taking all reasonable actions necessary to cure any defects in the proposed transfer, so that the Transfer of such LCFS Credits can be recorded. Each Party shall notify the other Party and the Executive Officer of any errors in the Credit Transfer Form within five (5) business days of the discovery of such an error.

3. REPRESENTATIONS AND WARRANTIES

3.1 Representations and Warranties by Both Parties

Each Party represents, warrants and covenants to the other Party as of the date of this Agreement (which representations and warranties are deemed to be repeated by each Party on each Transfer Date with respect to a Transaction) that:

- it is duly organized and validly existing under the laws of the jurisdiction of its organization or incorporation and, if relevant under such laws, in good standing;
- it has, and at all times during the term of this Agreement, will have, all necessary power and authority to execute, deliver, and perform its obligations hereunder;
- the execution, delivery, and performance of this Agreement by such Party has been duly authorized by all necessary action and do not violate any of the terms or conditions of its governing documents, any contract to which it is a party, or any Applicable Law;
- there is no pending or, to such Party's knowledge, threatened litigation or administrative proceeding that may materially adversely affect its ability to perform this Agreement;
- this Agreement constitutes a legal, valid and binding obligation of such Party, except as the enforceability of this Agreement may be limited by the effect of any applicable bankruptcy, insolvency, reorganization, moratorium or similar laws affecting creditors' rights generally and by general principles of equity; and
- Seller and Buyer shall comply with Applicable Law in the performance of their respective obligations under this Agreement and each (f) Transaction.

Representations and Warranties by Seller 3.2

Seller represents and warrants to Buyer that on each Transfer Date:

- Seller conveys good title to all LCFS Credits it sells hereunder, free and clear of any liens, security interests, and encumbrances or any interest in or to them by any third party;
- each LCFS Credit transferred to Buyer hereunder is valid as contemplated by the LCFS Regulations and is, indefeasibly, a "Credit" as defined by the LCFS Regulations;
- each LCFS Credit was deposited into Seller's LCFS Account, or otherwise transferred and recorded by the Executive Officer in Seller's LCFS Account, prior to Initiation hereunder and Seller has good title to each LCFS Credit prior to Initiation hereunder;
- the Quantity of LCFS Credits Initiated does not exceed the number of total LCFS Credits in the Seller's LCFS Account as determined in accordance with §95487(c)(1)(A) of the LCFS Regulations;
- upon Transfer and recordation of the LCFS Credits in Buyer's LCFS Account, the LCFS Credits shall be available for Buyer's use for retirement, transfer to a third party or otherwise;
- Seller has not sold, transferred or encumbered (nor become legally obligated to do the same) any rights, title, or interest in the LCFS Credits to any person other than Buyer; and
- neither the Seller, nor any of its associated or parent organizations or affiliates or its customers or the party that owns the project(s) producing the fuel that is the basis for the generation of the LCFS Credits, has claimed (or will be entitled to claim) directly or indirectly, including on any voluntary or mandatory greenhouse gas registry program, any of the LCFS Credits as anything other than sold to Buyer.

OTHER THAN THE WARRANTIES AND REPRESENTATIONS EXPRESSLY SET FORTH IN THIS AGREEMENT. TO THE EXTENT PERMITTED UNDER APPLICABLE LAW, SELLER MAKES NO OTHER REPRESENTATION OR WARRANTY, WRITTEN OR ORAL, EXPRESS OR IMPLIED, INCLUDING, WITHOUT LIMITATION, ANY REPRESENTATION OR WARRANTY WITH RESPECT TO MERCHANTABILITY OR FITNESS FOR ANY PARTICULAR PURPOSE.

PAYMENTS AND INVOICES 4.

4.1 **Trade Documentation**

Payee (generally Seller) shall promptly send payer (generally Buyer) a written invoice ("Invoice") showing sufficient detail to determine the amount due, how such amount was calculated, and the Payment Due Date for the Transaction in question, as set forth in the sample forms attached as Exhibit 2.

4.2 Making Payment

All payments shall be made in U.S. Dollars by the Payment Due Date via wire transfer in same day funds. Except as provided herein or as otherwise agreed by the Parties, payment shall be made without deduction, withholding or setoff. If a Payment Due Date falls on a Saturday or a day that is neither a New York Banking Day nor a Monday, payment shall be made on the preceding New York Banking Day. If a Payment Due Date falls on a Sunday or Monday that is not a New York Banking Day, payment shall be made on the following New York Banking Day.

4.3 Price Rounding

All U.S. dollar amounts shall be rounded to the nearest cent (and half cents shall be rounded upward).

4.4 Interest

Any amount payable hereunder, if not paid when due, and any amount payable as a refund as a result of an overpayment, shall bear interest from the Payment Due Date or the date of overpayment (as applicable) until the date payment is received at an annual rate (based upon the actual number of days in the relevant calendar year) equal to the rate of two (2) percentage points above the prime rate of interest effective for the Payment Due Date as published in the *Wall Street Journal* under "Money Rates". If there is no publication on the Payment Due Date, then the preceding day's publication will be used. The relevant Party shall pay any interest due within three (3) New York Banking Days following receipt of the interest invoice.

4.5 Payment Dispute

If a Party, in good faith, disputes the accuracy of the amount due in respect of a Transaction, such Party will timely pay such amount as it believes to be correct and provide written notice stating the reasons why the remaining disputed amount is incorrect, along with supporting documentation acceptable in industry practice. Payment of the disputed amount shall not be required until the dispute is resolved. In the event the Parties are unable to resolve such dispute, either Party may pursue any remedy available at law or in equity to enforce its rights pursuant to this Agreement. In the event that it is determined that the Party that is disputing the amount due must pay the disputed amount, then such Party shall pay interest in accordance with Section 4.4 on such disputed amount from and including the originally scheduled Payment Due Date to, but excluding the date paid.

5. <u>CONCURRENT TRANSACTIONS - PAYMENT NETTING</u>

5.1 Payment Netting

The Parties agree to net amounts that are due to each other on the same Payment Due Date in respect of purchases and sales of LCFS Credits, and the Parties shall confirm at least two (2) New York Banking Days prior to the Payment Due Date, orally or in writing, the payment obligations owed in respect of purchases and sales of LCFS Credits and any and all amounts remaining due after net-out. Any remaining balance after net-out shall be paid by the Party owing such amount to the other Party on the applicable Payment Due Date. The owing Party's payment of a net amount shall satisfy each Party's payment obligations under the relevant Transactions in respect of the payment obligations in respect of purchases and sales of LCFS Credits included in the settlement on a Payment Due Date. The Parties understand and agree that such netting of payment obligations is expressly limited to amounts owed from purchases and sales of LCFS Credits between the Parties and that netting out any other amounts due in respect of any Transaction or any Other Trading Agreement or any other agreement, instrument or undertaking, for any reason whatsoever, including amounts due in respect of interest payments, is strictly prohibited unless otherwise agreed in writing.

6. CREDIT

- 6.1 Notwithstanding any other Applicable Law, including the Uniform Commercial Code, if a Party (the "Secured Party") has commercially reasonable grounds for insecurity with respect to the other Party's (the "Posting Party") creditworthiness or performance under this Agreement, the Secured Party shall provide the Posting Party with written notice requesting an amount of Performance Assurance determined by the Secured Party in a commercially reasonable manner. "Performance Assurance" means either:
 - (a) prepayment, received by the Secured Party no later than two (2) New York Banking Days after such notice, and in any event
 prior to commencing the delivery period; or
 - (b) establishing, at the Posting Party's cost, no later than two (2) New York Banking Days after such notice an irrevocable standby letter of credit in a form and term acceptable to the Secured Party, opened by a Qualified Institution. "Qualified Institution" means either:
 - (i) a commercial bank, unaffiliated with either Party, organized in a jurisdiction reasonably acceptable to the Secured Party, that has:

- (A) at least an A- Long Term Rating issued by Standard & Poor's Ratings Group ("S&P") and at least an A3 Deposit Rating issued by Moody's Investor Services, Inc. ("Moody's"); and
- (B) total equity of at least ten billion U.S. dollars (\$10,000,000,000);
- (C) not exceeded any internal credit limits as established by the Secured Party at the time of the establishment of the letter of credit; or
- (ii) a bank acceptable to the Secured Party in its sole discretion; or
- (c) other Performance Assurance acceptable to the Secured Party in its sole discretion.
- 6.2 Notwithstanding anything to the contrary in this Agreement, if required by Seller, pursuant to Section 6.1 or as a result of a failure to pay or Event of Default, Buyer shall make advance payment in U.S. Dollars by electronic funds transfer to Seller for LCFS Credits purchased by Buyer pursuant to one or more Transactions under this Agreement ("Prepayment"). Specifically, for each Transaction pursuant to which Seller is obligated to transfer LCFS Credits to Buyer and for which Seller requires Buyer to make a Prepayment. Seller shall issue an invoice to Buyer and Buyer shall make the Prepayment to Seller by the date specified on Seller's invoice ("Prepayment Due Date"). All Prepayments shall be in an amount equal to the price (or estimated price if the price is based on an index and is not known at the time the invoice is issued) multiplied by the total quantity (or estimated quantity if the actual quantity is not known at the time the invoice is issued) of LCFS Credits to be purchased for each outstanding Transaction for which Prepayment is required (each a "Prepayment Amount"). Each Prepayment Amount shall be paid by electronic funds transfer, in same day funds (without setoff, counterclaim or deduction), to the account specified by Seller. If, pursuant to a Transaction, the actual quantity of LCFS Credits transferred differs from the contract quantity or the price differs from the estimated price upon which Prepayment was made or other amounts are owing by or to Buyer (including, without limitation, other charges related to the Transaction(s) or amounts arising from any overpayments or underpayments for prior periods), the Party owing such amounts shall pay such amounts owing by it within two (2) Business Days of receipt of request by the Party to whom the payment is owed. In the event Buyer fails to timely make the Prepayment, Seller shall have the right to immediately withhold or suspend transfer of LCFS Credits until such time as the required payment is received. Such suspension of transfer of LCFS Credits shall not relieve Buyer of its obligation to purchase LCFS Credits pursuant to any Transaction and shall be in addition to, and not in replacement of, any other right or remedy available to Seller under this Agreement.

7. REMEDIES FOR FAILURE TO INITIATE OR ACCEPT LCFS CREDITS, AND DEFICIENT LCFS CREDITS

- 7.1 In the event that, in relation to a Transaction:
 - (a) Seller fails to Initiate all or a part of the LCFS Credits by the end of the Transfer Period;
 - (b) Buyer exercises its right to reject all or part of a Quantity of LCFS Credits pursuant to Section 2.4 that Seller Initiated;
 - (c) Seller breaches any of its representations or warranties in Section 3.2, or any representation or warranty specified as subject to this Section 7 in the applicable Confirmation;
 - (d) LCFS Credits that have been Accepted by Buyer are or become invalid for purposes of the LCFS Regulations; or
 - (e) the Executive Officer invalidates a Transfer of LCFS Credits from Seller to Buyer, for any reason, and Buyer and Seller, cooperating in good faith, are unable to remedy the invalidated Transfer.

(each such affected LCFS Credit, a "<u>Deficient LCFS Credit</u>"), then, Seller shall, at Seller's sole cost and expense, Initiate a quantity of Qualified Replacement LCFS Credits equal to the quantity of Deficient LCFS Credits that satisfy the requirements under the applicable Transaction within (i) in the case of Sections 7.1(a) or (b), three (3) Business Days after Seller receives notice from Buyer that the circumstances in Section 7.1(a) or (b) apply, or (ii) in the case of Sections 7.1(c), (d) or (e), ten (10) Business Days after the earlier of (A) Seller's receipt of notice from Buyer that the circumstances in Sections 7.1(c), (d) or (e) apply and (B) Seller's becoming aware that the circumstances in Sections 7.1(c), (d) or (e) apply.

7.2 Except where Section 7.3 applies, if Seller fails to timely or fully comply with its Initiation obligation in Section 7.1, then Seller shall, at Buyer's election by notice either (i) Initiate replacement LCFS Credits equal to the quantity of Deficient LCFS Credits that satisfy the requirements under the applicable Transaction in accordance with Section 7.1 above, or (ii) pay to Buyer, within five (5) Business Days of receipt of Buyer's invoice, unless otherwise mutually agreed between the Parties, the positive difference, if any, between (1) the Market Value of the Qualified Replacement LCFS Credits and (2) the product of the quantity of Deficient LCFS Credits and the Contract Price specified in the Confirmation for the applicable LCFS Credits, with such sum increased by any amount already paid by Buyer to Seller on account of the Deficient LCFS Credits. Seller shall also reimburse Buyer for all reasonable broker costs associated with Buyer obtaining Qualified Replacement LCFS Credits in a quantity equal to the Deficient LCFS Credits.

For purposes of this Section 7.2, (i) the phrase "the quantity of LCFS Credits that remain to be Initiated under that Transaction" as used in the defined terms "<u>Market Value</u>" shall refer to and mean the quantity of Deficient LCFS Credits; and (ii) the Market Value shall be determined by Buyer as of the Business Day notified by Buyer that falls no sooner than the last day for performance of Seller's obligations under Section 7.1 and no later than three (3) Business Days after the date Buyer gives notice of its election.

7.3 In the event that Buyer fails to Accept, or provide notice of its rejection of, all or part of a Quantity of LCFS Credits Delivered by Seller or

any replacement LCFS Credits as contemplated under Section 2.4, Seller shall provide written notice of such failure to Buyer. If Buyer fails to Accept or reject any portion of those LCFS Credits (the "Pending Credits") within two (2) Business Days of receiving such notice, then Seller's obligation to sell and Initiate and Buyer's obligation to purchase and Accept shall be reduced to the extent of such Pending Credits, and Buyer shall pay Seller an amount equal to the positive difference, if any, between (i) the product of the quantity of Pending Credits and the Contract Price specified in the Confirmation for the applicable LCFS Credits and (ii)the Market Value of the Pending Credits. The Transaction in respect of the Pending Credits shall be deemed to be cancelled and the related Credit Transfer Form or LRT Transfer Notification, as applicable, shall be deemed ineffective. For purposes of this Section 7.3, Market Value shall be determined by Seller in a commercially reasonable manner with the date of determination as of the date of cancellation.

- 7.4 In the event the provisions of this Section 7 are invoked, Seller and Buyer agree to work together in good faith to pursue an efficient, commercial and practical resolution consistent with the foregoing options (or any combination thereof) in order to cure any breach of transfer obligations resulting in Deficient LCFS Credits or Pending Credits, provided, however, any replacement LCFS Credits must satisfy all of the requirements that the Parties originally agreed to for the applicable Transaction unless otherwise mutually agreed.
- 7.5 Seller shall deliver to Buyer a Credit Transfer Form or LRT Transfer Notification accurately describing any Qualified Replacement LCFS Credits. Buyer and Seller shall otherwise be subject to the general obligations set forth in Section 2.
- 7.6 Section 7.1(b), (c), (d), (e) and, for the avoidance of doubt, Section 2.4 shall apply equally to any Qualified Replacement LCFS Credits.
- 7.7 Except in respect of a failure to pay any amount due under Section 7.2 or Section 7.3, the remedies set out in this Section 7 are exclusive remedies for the occurrence of the events described in Section 7.

8. EVENTS OF DEFAULT

- 8.1 An "Event of Default" shall mean the occurrence with respect to a Party of one of the following events:
 - (a) <u>Failure to Pay</u>. A Party fails to make payment of any amount due when required under this Agreement or any Transaction, within two (2) New York Banking Days following receipt of a written notice of such failure from the other Party.
 - (b) <u>Failure to Provide Performance Assurance</u>. A Party fails to provide acceptable Performance Assurance support as requested by the Secured Party pursuant to Section 6.
 - (c) <u>Breach of Agreement</u>. Except for any breach or event described in Section 7.1(a) through (e) (the exclusive remedies for which are specified in Section 7.2) or Section 7.3, and except for any event described in Section 8.1(a) and (b) above, a Party fails to perform or repudiates any material obligation to the other Party under this Agreement or breaches any representation, covenant or warranty in any material respect under this Agreement and, in each case, if capable of being cured, is not cured to the satisfaction of the other Party in its sole discretion, within two (2) New York Banking Days following receipt of written notice to such Party that corrective action is needed.
 - (d) <u>Default Under Other Trading Agreements</u>. If specified as applicable in the Cover Sheet, a Party, any Credit Support Provider of such Party, or any applicable Specified Affiliate of such Party.
 - (i) fails to make payment when due under any trading agreement (that is not a Transaction under this Agreement) with the other Party, any Credit Support Provider of such other Party or any Specified Affiliate of such other Party, including any agreement to purchase, sell or exchange LCFS Credits or any other commodities or any agreement in respect of an interest rate transaction, equity transaction, bond transaction, foreign exchange transaction, credit protection transaction, repurchase transaction, buy/sell-back transaction, securities lending transaction, forward transaction, any transaction similar to the foregoing, any swap, forward, future, option or other derivative transaction with respect to the foregoing or any combination of these transactions (each, an "Other Trading Agreement") within the cure period for payment defaults specified therein, or if no period is provided, within two (2) New York Banking Days following receipt of a demand for payment by the other Party; or
 - (ii) commits or suffers an Event of Default or other similar event or condition, fails to perform or repudiates any obligation to the other Party, any Credit Support Provider of such other Party or any Specified Affiliate of such other Party under any Other Trading Agreement, or breaches any representation, covenant or warranty in any material respect under any Other Trading Agreement;

and, in the case of either of subclauses (i) or (ii) above, (1) such failure, breach or default results in a liquidation of, an acceleration of obligations under, or an early termination of, all transactions outstanding under the documentation applicable to such Other Trading Agreement; and (2) the aggregate amount owed by the Defaulting Party in respect of the liquidation of, acceleration of obligations under or early termination of all transactions outstanding under the documentation applicable to such Other Trading Agreement(s) is not less than the Aggregate Delinquency Amount applicable to such Party (it being agreed that the failure to make a payment or delivery the value of which exceeds the Aggregate Delinquency Amount on the last payment or delivery date of an Other Trading Agreement transaction where that transaction is the only transaction outstanding under the applicable documentation will satisfy conditions (1) and (2)).

(e) <u>Cross Default.</u> A default, Event of Default or other similar condition or event occurs and is continuing in respect of a Party or its Credit Support Provider (if any) under one or more agreements or instruments, individually or collectively, relating to indebtedness for borrowed money in an aggregate amount of not less than the Threshold Amount applicable to such Party and which results in such indebtedness becoming, or becoming capable at such time of being declared, immediately due and payable; or (ii) a Party or its

- Credit Support Provider (if any) fails to make on the due date thereof one or more payments, individually or collectively, in an aggregate amount of not less than the Threshold Amount applicable to such Party.
- (f) <u>Designated Event</u>. A Designated Event occurs with respect to a Party or its Credit Support Provider (if any), and the creditworthiness of the Party or its Credit Support Provider or, if applicable, the successor, surviving or transferee entity of the Party or its Credit Support Provider (as applicable) is materially weaker than that of the Party or its Credit Support Provider immediately prior to such Designated Event.
- (g) Credit Support Failure. A Party's guarantor or other provider of credit support for such Party (a "Credit Support Provider") with respect to the Party, if any, (i) fails to satisfy, perform or comply with any agreement or obligation to be complied with or performed by it in accordance with the credit support document and such failure continues after any applicable grace or notice period, (ii) makes any representation or warranty that proves to be incorrect or misleading in any material respect when made in connection with such Performance Assurance and/or credit support document, or (iii) repudiates, disclaims, disaffirms or rejects, in whole or part, any obligation under, or challenges the validity of, its Performance Assurance and/or credit support document.
- Bankruptcy. A Party or its Credit Support Provider, if any, is or becomes Bankrupt.
- (i) Merger Without Assumption. If a Designated Event (as such term is defined in 1.1(u) and notwithstanding the reference to Section 8.1(f) therein) occurs with respect to a Party or any Credit Support Provider of such Party and:
 - (i) the resulting, surviving or transferee entity fails to assume all the obligations of such Party or such Credit Support Provider under this Agreement or any credit support document to which it or its predecessor was a party by operation of law or pursuant to an agreement reasonably satisfactory to the other Party to this Agreement; or
 - (ii) the benefits of any credit support document fail to extend (without the consent of the other Party) to the performance by such resulting, surviving or transferee entity of its obligations under this Agreement.

9. TERMINATION AND LIQUIDATION

- 9.1 Notwithstanding any other provision of this Agreement or the existence of any Performance Assurance, if at any time an Event of Default has occurred and is continuing with respect to a Party (such Party, the "<u>Defaulting Party</u>"), the other Party (the "<u>Performing Party</u>") may, in its sole discretion, designate a date (not earlier than the date of such notice and not later than twenty (20) days after the date of such notice (an "<u>Early Termination Date</u>")) on which to terminate, liquidate and accelerate all outstanding Transactions and calculate a Termination Payment (as defined below) in the manner set forth in Section 9.2 and Section 9.3. To the extent that, in the reasonable opinion of the Performing Party, certain Transactions may not be liquidated and terminated under Applicable Law on the Early Termination Date, such Transactions shall be terminated as soon thereafter as is reasonably practicable, in which case the actual termination date for such Transactions will be the Early Termination Date in respect thereof for purposes of Section 9.2. Notwithstanding the foregoing, if the Defaulting Party is governed by a system of law that does not permit termination to take place after the occurrence of an Event of Default described in Section 8.1(h), then no prior notice shall be required upon the occurrence of such Event of Default, in which case the Early Termination Date shall be deemed designated immediately preceding the occurrence of such event.
- On or as soon as reasonably practicable following the Early Termination Date, the Performing Party shall determine the final amount 9.2 payable between the Parties under this Agreement as provided in this Section 9.2 (the "Termination Payment") and shall provide notice of the Termination Payment to the Defaulting Party. The Performing Party shall calculate the Termination Payment by (a) valuing each Transaction at its Market Value as reasonably determined by the Performing Party as of the Early Termination Date and then determining the amount by which such then prevailing Market Value differs from the Contract Value (it being understood that (i) in the event the prevailing Market Value of a Transaction exceeds the Contract Value, the difference in value shall be due from Seller to Buyer, and (ii) in the event that the prevailing Market Value of a Transaction is less than the Contract Value, the difference in value shall be due from Buyer to Seller), (b) determining any other damages, costs or expenses incurred by the Performing Party as a result of the early termination of such Transactions including those contemplated by Section 9.4 (without duplication and subject always to Section 13.1), (c) determining any other amounts payable from one Party to the other Party under this Agreement (including amounts due in respect of LCFS Credits Initiated and Accepted hereunder) and (d) netting or aggregating the foregoing amounts into a single liquidated amount. If the Defaulting Party owes the Termination Payment to the Performing Party, then, within one (1) New York Banking Day of the date upon which the Performing Party's notice of the Termination Payment is effective, the Defaulting Party shall pay the Termination Payment, less the value of any Performance Assurance or other collateral or credit support held by the Performing Party with respect to which the Performing Party has notified the Defaulting Party in writing of its election to exercise its setoff rights under Section 9.3. If the Performing Party owes the Termination Payment to the Defaulting Party, then, within one (1) New York Banking Day of the date upon which the Performing Party's notice of the Termination Payment is effective, the Performing Party shall pay the Termination Payment, less the value of any Performance Assurance or other collateral or credit support held and not returned by the Defaulting Party.

9.3 Closeout Setoff

Option A (Bilateral Setoff)

If no option or Option A is specified as applicable in the Cover Sheet, then the following provision shall apply:

If the Performing Party elects to designate an Early Termination Date under Section 9.1, and the Termination Payment is payable to the Defaulting Party, the Performing Party shall be entitled, at its option and in its discretion (and without prior notice to the Defaulting Party), to setoff against such Termination Payment any amounts ("Other Amounts") payable by the Defaulting Party to the Performing Party under any other agreements, instruments or undertakings between the Defaulting Party and the Performing Party (whether or not arising under this Agreement, matured or contingent and irrespective of the currency, place of payment or place of booking of the obligation). To the extent that any Other Amounts are so setoff, those Other Amounts will be discharged promptly and in all respects. The Performing Party will give notice to the other Party of any setoff effected under this Section 9.3. For this purpose, either the Termination Payment or the Other Amounts (or the relevant portion of such amounts) may be converted by the Performing Party into the currency in which the other is denominated at the rate of exchange at which the Performing Party would be able, in good faith and using commercially reasonable procedures, to purchase the relevant amount of such currency.

Option B (Triangular Setoff)

If Option B is specified as applicable in the Cover Sheet, then the following provision shall apply:

If the Performing Party elects to designate an Early Termination Date under Section 9.1, and the Termination Payment is payable to the Defaulting Party, the Performing Party shall be entitled, at its option and in its discretion (and without prior notice to the Defaulting Party), to setoff against such Termination Payment any amounts ("Other Amounts") payable by the Defaulting Party to the Performing Party or any of the Performing Party's Affiliates under any other agreements, instruments or undertakings between the Defaulting Party and the Performing Party or any of its Affiliates (whether or not arising under this Agreement, matured or contingent and irrespective of the currency, place of payment or place of booking of the obligation). To the extent that any Other Amounts are so setoff, those Other Amounts will be discharged promptly and in all respects. The Performing Party will give notice to the other Party of any setoff effected under this Section 9.3. For this purpose, either the Termination Payment or the Other Amounts (or the relevant portion of such amounts) may be converted by the Performing Party into the currency in which the other is denominated at the rate of exchange at which the Performing Party would be able, in good faith and using commercially reasonable procedures, to purchase the relevant amount of such currency.

Option C (Rectangular Setoff)

If Option C is specified as applicable in the Cover Sheet, then the following provision shall apply:

If the Performing Party elects to designate an Early Termination Date under Section 8.1, the Performing Party shall be entitled, at its option and in its discretion (and without prior notice to the Defaulting Party), to setoff against such Termination Payment (whether such Termination Payment is payable to the Performing Party or to the Defaulting Party) any amounts ("Other Amounts") payable between the Defaulting Party or any of its Affiliates and the Performing Party or any of its Affiliates under any other agreements, instruments or undertakings between the Defaulting Party or any of its Affiliates to the Performing Party or any of its Affiliates (whether or not arising under this Agreement, matured or contingent and irrespective of the currency, place of payment or place of booking of the obligation). To the extent that any Other Amounts are so setoff, those Other Amounts will be discharged promptly and in all respects. The Performing Party will give notice to the other Party of any setoff effected under this Section 9.3. For this purpose, either the Termination Payment or the Other Amounts (or the relevant portion of such amounts) may be converted by the Performing Party into the currency in which the other is denominated at the rate of exchange at which the Performing Party would be able, in good faith and using commercially reasonable procedures, to purchase the relevant amount of such currency.

Option D (No Setoff)

If Option D is specified as applicable in the Cover Sheet, then no setoff provisions shall apply.

No delay or failure on the part of a Performing Party to exercise any right or remedy shall constitute an abandonment of such right or 9.4 remedy, and the Performing Party shall be entitled to exercise such right or remedy at any time after an Event of Default has occurred, so long as such Event of Default is continuing. The Defaulting Party shall indemnify and hold harmless the Performing Party for and against any and all reasonable out-of-pocket expenses incurred by the Performing Party by reason of the enforcement of and protection of its rights under this Agreement or as a result of the early termination of any Transactions, including reasonable attorney's fees and costs of collection.

Grant of Security Interest/Remedies 9.5

To the extent a Party requires Performance Assurance and/or has received a credit support document under this Agreement, then the Posting Party hereby grants to the Secured Party a present and continuing security interest in same. Upon or at any time after the designation or deemed designation of an Early Termination Date, the Defaulting Party must return all remaining Performance Assurance transferred to it pursuant to this Agreement, as applicable, and all proceeds resulting therefrom or the liquidation thereof.

9.6 Suspension of Performance

Notwithstanding any other provision of this Agreement, if (a) an Event of Default or (b) an event that, with the lapse of time or the giving of notice or both, would constitute an Event of Default shall have occurred and be continuing, the Performing Party, upon written notice to the Defaulting Party, shall have the right (i) to suspend performance under any or all Transactions; provided, however, in no event shall any such suspension continue for longer than ten (10) New York Banking Days with respect to any single Transaction, unless an Early Termination Date shall have been declared and notice thereof given pursuant to Section 9.1, and (ii) to the extent an Event of Default shall have occurred and be continuing, to exercise any remedy available at law or in equity.

9.7 **Bankruptcy Acknowledgements**

The Parties intend that each Transaction shall constitute a "forward contract" under § 101(25) and a swap agreement under § 101(53b) of Title 11 of the United States Code, 11 U.S.C. § 101 et seq., as amended from time to time (the "Bankruptcy Code"), and that this Agreement constitutes a "master netting agreement" under § 101(38a) of the Bankruptcy Code, and that the rights of the Performing Party in Section 9 include the rights referred to in § 561(a) of the Bankruptcy Code. Further, the Parties intend that each Party shall be a "forward contract merchant" under § 101(26) and a "master netting agreement participant" under § 101(38B), for purposes of the Bankruptcy Code.

10. **FORCE MAJEURE**

- 10.1 Subject to Section 10.2, a Party shall be excused from the performance of its obligations with respect to a Transaction to the extent its performance of such obligations is prevented, in whole or in part, due to the occurrence of any event or circumstance, whether foreseeable or unforeseeable, that is reasonably beyond the control of such Party and which, by the exercise of due diligence, such Party could not have remedied, avoided or overcome (any such event, a "Force Majeure"), which may include, without limitation, any of the following events:
 - Compliance with Applicable Law, provided however, that Seller shall not be excused from performance where the LCFS Credits it Initiates or intends to Initiate are invalid under the LCFS Regulations;
 - Hostilities of war (declared or undeclared), embargoes, blockades, civil unrest, riots or disorders, acts of terrorism, or sabotage;
 - Fires, explosions, lightning, maritime peril, collisions, storms, landslides, earthquakes, floods, and other acts of nature;
 - Strikes, lockouts, or other labor difficulties (whether or not involving employees of Seller or Buyer); provided, however, that the decision to settle a strike or other labor difficulties shall be wholly within the discretion of the Party facing such difficulty; or
 - Disruption or breakdown of production or transportation facilities, equipment, labor or materials, including, without limitation, the closing of harbors, railroads or pipelines.

For purposes of this Agreement, the term "Force Majeure" expressly excludes (i) a failure of performance of any person other than the Parties, except to the extent that such failure was caused by an event that would otherwise satisfy the definition of a Force Majeure event as set forth in this Section 10, (ii) the loss of Buyer's market or any market conditions for any LCFS Credits that are unfavorable for Buyer or Seller, (iii) the loss of Seller's intended supply of LCFS Credits, (iv) the failure of Seller's intended supplier of LCFS Credits to perform, (v) any failure by a Party to apply for, obtain or maintain any permit, license, approval or right of way necessary under Applicable Law for the performance of any obligation hereunder, and (vi) a Party's inability to economically perform its obligations under this Agreement.

LRT-CBTS Unavailability. In the event that LRT-CBTS is disrupted or unavailable, the affected obligations of the parties will be suspended (but not discharged) until LRT-CBTS is not disrupted and is available.

- 10.2 Notwithstanding the provisions of Section 10.1, nothing contained in this Agreement shall relieve a Party of its obligation to make payments when due with respect to performance prior to the occurrence of a Force Majeure event, including Buyer's obligation to pay in full the purchase price or any other amounts due for the LCFS Credits actually Initiated and Accepted hereunder.
- In the event that a Party believes a Force Majeure event has occurred that will require it to invoke the provisions in this Section 10, such 10.3 Party shall use commercially reasonable efforts to give prompt oral notice to the other Party followed by written notice within two (2) New York Banking Days following the occurrence of such event, of the underlying circumstances of the particular causes of Force Majeure, the expected duration thereof and the volume of the LCFS Credits affected. The Party claiming Force Majeure shall also use commercially reasonable efforts to give the other Party such notice of cessation of the Force Majeure event and the date when performance is expected to resume.
- 10.4 Notwithstanding anything to the contrary in this Agreement, (a) if an event or circumstance which would otherwise constitute or give rise to a Force Majeure event under this Section 10 also constitutes an Event of Default other than an Event of Default under Section 8.1(h), it will be treated as a Force Majeure event and not as an Event of Default; and (b) if an event or circumstance which would otherwise constitute or give rise to a Force Majeure event under this Section 10 also constitutes an Event of Default under Section 8.1(h), it will be treated as an Event of Default and not as a Force Majeure event.
- 10.5 If Initiations and Acceptances in respect of a Term Transaction are suspended pursuant to this Section 10 and said suspension continues for a period of thirty (30) days or more, such Term Transaction may be terminated at the option of (a) either Party, if the "Force Majeure Termination Payment" provision is specified as applicable in the Cover Sheet, or (b) the Party that is not claiming Force Majeure, if the Force Majeure Termination Payment provision is not specified as applicable in the Cover Sheet, in either case, by giving written notice to the other Party. If a Party terminates a Term Transaction pursuant to this Section 10.5, neither Party shall have any further liability to the other Party hereunder with respect to such Term Transaction except for (a) any payment or indemnification obligations arising in respect of the performance of such Term Transaction prior to the date of termination and (b) if the Force Majeure Termination Payment provision is specified as applicable in the Cover Sheet, the obligation to make a Force Majeure Termination Payment, if any, pursuant to such

provision.

11. GOVERNING LAW AND SETTLEMENT OF DISPUTES

- This Agreement shall be governed by and construed in accordance with the laws of the State of New York without reference to its choice of law doctrine, but without prejudice to the provisions of § 5-1401 of the General Obligations Law of the State of New York. The Parties hereby submit to the exclusive jurisdiction of any federal court of competent jurisdiction, or, if any federal court declines to exercise or does not have jurisdiction, in any New York state court situated in New York City, Borough of Manhattan, and to service of process by certified mail delivered to the Party at its last designated address. Each Party waives, to the fullest extent permitted by Applicable Law, any right it may have to a trial by jury in respect of any proceedings relating to this Agreement.
- 11.2 The Parties expressly agree that the United Nations Convention on Contracts for the International Sale of Goods shall not apply to this Agreement.

12. TAXES

12.1 Tax Obligations Generally

Each Party shall be responsible for any taxes that may be imposed on it arising from the sale or purchase, respectively, of LCFS Credits pursuant to any Transaction.

13. LIMITATION OF LIABILITY

NO PARTY SHALL BE REQUIRED TO PAY OR BE LIABLE FOR INCIDENTAL, CONSEQUENTIAL, INDIRECT, EXEMPLARY, SPECIAL OR PUNITIVE DAMAGES OR FOR LOST PROFITS OR ANY FINES OR PENALTIES ASSESSED BY ANY GOVERNMENTAL AUTHORITY INCLUDING, BUT NOT LIMITED TO, FINES OR PENALTIES UNDER THE LCFS REGULATIONS (WHETHER OR NOT ARISING FROM ITS NEGLIGENCE) TO ANY OTHER PARTY EXCEPT TO THE EXTENT THAT THE PAYMENTS REQUIRED TO BE MADE PURSUANT TO THIS AGREEMENT (INCLUDING PAYMENTS REQUIRED TO BE MADE PURSUANT TO SECTIONS 7, 9 AND 19) ARE DEEMED TO BE SUCH DAMAGES. IF AND TO THE EXTENT ANY PAYMENT REQUIRED TO BE MADE PURSUANT TO THIS AGREEMENT IS DEEMED TO CONSTITUTE LIQUIDATED DAMAGES, THE PARTIES ACKNOWLEDGE AND AGREE THAT SUCH DAMAGES ARE DIFFICULT OR IMPOSSIBLE TO DETERMINE AND THAT SUCH PAYMENT IS INTENDED TO BE A REASONABLE APPROXIMATION OF THE AMOUNT OF SUCH DAMAGES AND NOT A PENALTY. EACH PARTY SHALL TAKE REASONABLE STEPS TO MITIGATE DAMAGES FROM ANY BREACH HEREOF.

14. ASSIGNMENT

- 14.1 This Agreement shall not be assigned, in whole or in part, by either Party without the prior written consent of the other Party, such consent not to be unreasonably withheld. Any assignment in violation of this provision shall be void and of no legal effect. Subject to this Section 14, this Agreement shall be binding upon and shall inure to the benefit of the Parties and their respective successors and assigns.
- 14.2 Notwithstanding anything to the contrary in Section 14.1, either Party may, without the other Party's consent, transfer, sell, pledge, encumber or assign such Party's right and interest in and to the accounts, revenues, or proceeds of this Agreement in connection with any financing or other financial arrangement. For the avoidance of doubt, all payment obligations under this Agreement will be satisfied if the relevant payment is made to the account of the other Party and the foregoing sentence shall not require a Party to make payments to any third party, unless otherwise agreed in writing.

15. NON-WAIVER

No waiver by either Party of any breach by the other Party of any of the representations, covenants, warranties, terms or conditions of this Agreement shall be construed as a waiver of any succeeding breach of the same or of any other covenant or condition hereof.

16. ENTIRE AGREEMENT; AMENDMENTS

No statement or agreement, oral or written, made prior to the signing of this Agreement, shall vary or modify the written terms hereof. Neither Party shall claim any amendment to, modification of, or release from any provisions of this Agreement, whether set out in an annex, schedule, supplement or Confirmation or otherwise, unless such amendment, modification or release is in writing, is signed by the Parties and specifically states that it is an amendment to, modification of, or release from this Agreement or one or more Transactions.

17. NOTICES

All notices and other communications under this Agreement shall be deemed given on the date of the addressee's receipt thereof and shall be given only in writing by letter, facsimile or electronic data transmission. Provided that if transmitted by facsimile or electronic data transmission and it is received after close of business on a New York Banking Day, it shall be deemed to have been received on the following New York Banking Day.

18. WRITTEN CONFIRMATIONS

- 18.1 The Parties intend that they are legally bound by the terms of each Transaction from the moment they agree to those terms (whether orally or otherwise). A Confirmation may be generated electronically by an electronic confirmation matching service or be executed and delivered in counterparts (including by facsimile transmission or by other means agreed between the Parties), which in each case will be sufficient for all purposes to evidence a binding supplement to this Agreement.
- Absent the availability of an electronic confirmation matching service, the Confirming Party shall confirm a Transaction by forwarding a written Confirmation to the other Party (via facsimile or other means agreed between the Parties) within five (5) New York Banking Days after the Trade Date. If the other Party objects to any term(s) of such written Confirmation, it shall notify the Confirming Party in writing of such objection within ten (10) calendar days of the Party's receipt thereof. If the other Party fails to object within ten (10) calendar days, such failure shall be deemed acceptance of the terms, absent manifest error.

19. NEW OR CHANGED LAWS OR REGULATIONS

19.1 Material Adverse Effect

If this Section 19.1 is specified as applicable in the Cover Sheet, then the following provision shall apply to Transactions under this Agreement:

It is understood by the Parties that each Party is entering into this Agreement in reliance on the Applicable Laws in effect on the date hereof. In the event that at any time, and from time to time, during the term of this Agreement any Applicable Laws are changed or new Applicable Laws are promulgated and have a material adverse economic effect upon either Party (such Party being the "Affected Party"), and such event does not constitute a Force Majeure, the Affected Party shall have the option (the "Renegotiation Option") to request renegotiation of the Price or other terms of the affected transaction(s) (the "Affected Transactions"). The Renegotiation Option may be exercised by the Affected Party within a commercially reasonable period of time after such changed or new Applicable Law is promulgated, by written notice of the Affected Party's desire to renegotiate, such notice to contain specific information about the new or changed Applicable Law and how it has a material adverse economic effect upon the Affected Party and the new Price or terms desired by the Affected Party to remedy or reverse such material adverse economic effect (the "Renegotiation Notice").

The issue as to whether the changed or new Applicable Law has a material adverse economic effect on the Affected Party shall be submitted promptly for resolution to an Expert, provided that the other Party shall have the option to agree that there has been a material adverse economic effect and waive its right to an Expert resolution at any time. The cost of the Expert shall be shared equally between the Parties. The Expert shall determine that the Applicable Law either has a material adverse economic effect or does not. The Expert's decision (the "Expert's Decision") shall be final and binding, absent fraud, and shall be provided to the Parties in the form of a report prepared by the Expert that contains facts and other data supporting the Expert's Decision. If the Expert's Decision concludes that the new or changed Applicable Law has a material adverse economic effect on the Affected Party, then the Parties shall in good faith renegotiate the Price or other terms in order to appropriately address the material adverse economic effects of the new or changed Applicable Laws.

If an Expert does not provide an Expert's Decision within thirty (30) days after receipt of the Renegotiation Notice by the other Party, or such other time period as agreed between the parties, the new or changed Applicable Law shall be deemed to have a material adverse economic effect on the Affected Party. If the Parties fail to agree upon a new Contract Price or terms satisfactory to both Parties within thirty (30) days after the other Party receives the Renegotiation Notice, or such other time period as agreed between the parties, the Affected Party shall have the right to terminate the Affected Transactions as of the end of the applicable thirty (30) day period (the "Termination Date") by giving written notice thereof to the other Party on or before such Termination Date in which case each Party shall make a determination of the Termination Value of the Affected Transactions within three (3) New York Banking Days of the Termination Date. The amount payable in settlement of the termination of the Affected Transactions (the "Settlement Amount") shall equal the sum of (i) one-half of the difference between the Termination Values calculated by the two Parties, and (ii) the lesser of such two Termination Values. If either Party disputes the resulting Settlement Amount as commercially unreasonable by written notice to the other Party within five (5) New York Banking Days of the Termination Date, each Party shall appoint one Independent Dealer to make a determination regarding whether the Settlement Amount is commercially reasonable. If both Independent Dealers agree that the Settlement Amount is commercially reasonable, then that value shall be final and binding and shall be paid promptly by the Party owing the Settlement Amount to the other Party. If one or more of the Independent Dealers determines that the Settlement Amount is not commercially reasonable, then each Independent Dealer will make an alternative determination of the Termination Value, and the amount payable between the Parties with respect to Affected Transactions (the "Alternative Settlement Amount") will be an amount equal to the sum of (i) one-half of the difference between the alternative Termination Values calculated by the two Independent Dealers, and (ii) the lesser of such two alternative Termination Values.

Each Party's and, if applicable, each Independent Dealer's, determination of the Termination Value shall not take into account changes to the economics of the Affected Transactions that would result from the new or changed Applicable Laws.

In the event that the first Independent Dealer approached by a Party refuses to act as Independent Dealer for purposes of dispute resolution hereunder, the Party will approach at least one additional Independent Dealer to assess whether it would be willing to act as Independent Dealer for purposes of dispute resolution hereunder. If either Party is unable to find an Independent Dealer willing to make a determination of the Termination Value for purposes of dispute resolution after approaching at least two Independent Dealers, then the originally determined Settlement Amount shall be deemed final and binding.

Upon the successful appointment of an Independent Dealer, each of the Parties may submit written support for the Transaction Value determined by such Party to the Independent Dealer. Copies of all information submitted by a Party to any Independent Dealer shall be provided to the other Party.

Any LCFS Credits Transferred within thirty (30) days of the Renegotiation Notice shall be provisionally invoiced and paid, but will be subject to adjustment based on the renegotiated price or terms, if agreement is reached on price or terms.

19.2 **Disrupted Transactions**

If this Section 19.2 is specified as applicable in the Schedule, then the following Provisions shall apply to Transactions under this Agreement:

If the price of a Transaction is based on an industry reference index (the "Original Index") that ceases to be published or is not published for any period applicable to calculation of the Reference Price of one or more Transactions (the "Disrupted Transactions"), the Parties shall in good faith (a) select an alternative index that reflects as nearly as possible the same information as published in the Original Index; or (b) negotiate an interim Reference Price for the Disrupted Transaction until the Original Index recommences publishing or an alternative index can be identified to replace the Original Index. In the event the Parties are unable to agree on an alternative index or otherwise agree on a price within ten (10) days after the Original Index ceases to be published or is not published for any period applicable to calculation of the Reference Price of a Disrupted Transaction, then the issue shall be promptly submitted to an Expert for resolution. The cost of the Expert shall be shared equally between the Parties. Upon appointment of the Expert, each Party shall submit to the Expert its determination of the Price for the Disrupted Transaction. The Expert shall select from the two submissions the Price that it determines best reflects the Price that would have applied to the Disrupted Transaction based on the Original Index in the absence of its disruption. The Expert's decision shall be final and binding, absent fraud, and shall be provided to the Parties in the form of a report that contains the facts and other data supporting the Expert's decision. The Expert's decision shall be provided to the Parties within thirty (30) days after the Original Index ceases to be published or is not published for any period applicable to calculation of the Reference Price of a Disrupted Transaction.

MISCELLANEOUS 20.

20.1 Severability

If any Governmental Authority of competent jurisdiction declares any provision of this Agreement unlawful, void or unenforceable, such provision will not invalidate, void or make unenforceable any other provision of this Agreement. The remaining terms and conditions shall remain in full force and effect, and the Parties will negotiate in good faith to reform this Agreement in order to give effect to the original intention of the Parties.

Recording of Conversations 20.2

Each Party hereby acknowledges to the other Party and consents that such other Party from time to time and without further notice and to the extent permitted by law:

- may record and retain electronic transmissions (including telephone conversations, e-mail and instant messaging) between the Parties' respective commercial marketing traders, marketers, schedulers and operations personnel in connection with this Agreement or any Transaction on central and local databases for their respective legitimate business purposes;
- may monitor electronic transmissions through their internal and external networks for purposes of security and compliance with Applicable Laws, regulations and internal policies for their other legitimate business purposes; and
- will comply with all Applicable Laws relating to the recording of electronic transmissions.

Without limiting the foregoing, each Party will notify all of its respective commercial marketing traders, marketers, schedulers and operations personnel who engage in these electronic transmissions of such recording and shall obtain their prior consent to such recording.

No Third Party Beneficiaries 20.3

Nothing expressed or implied in this Agreement is intended to create any rights, interests, obligations or benefits under this Agreement in any person other than the Parties and their respective successors and permitted assigns.

20.4 **Ethics**

Each Party warrants that neither the Party nor its directors, employees, agents, or subcontractors have given commissions, rebates, payments, kickbacks, or lavish or expensive gifts, entertainments or other things of value (individually and collectively referred to as 'Inappropriate Payments") to any director, employee, agent, or subcontractor of the other Party in connection with the Agreement and acknowledges that the giving of Inappropriate Payments may result in the cancellation of this and all other future Agreements. Each Party will notify the other Party of any such Inappropriate Payments by any of its directors, employees, agent or subcontractors.

Counterparts 20.5

This Agreement (including any Confirmation) may be executed in any number of counterparts and by different Parties in separate counterparts, any of which when so executed shall be deemed to be an original and all of which when taken together shall constitute one and the same agreement.

20.6 Confidentiality

The contents of this Agreement, any Transaction, and all other documents relating to this Agreement, and any information made available by one Party or its Credit Support Provider to the other Party or its Credit Support Provider with respect to this Agreement is confidential and shall not be disclosed to any third party (nor shall any public announcement relating to this Agreement be made by either Party), except for such information:

- (a) as may become generally available to the public:
- (b) as may be required or appropriate in response to any summons, subpoena, or otherwise in connection with any litigation or to comply with any applicable law, order, regulation, ruling, or accounting disclosure rule or standard;
- (c) as may be obtained from a non-confidential source that disclosed such information in a manner that did not violate its obligations to the non-disclosing Party or its Credit Support Provider;
- (d) to the extent necessary to comply with a Governmental Authority's request or reporting requirements, but both Parties shall submit all information to a Governmental Authority as confidential business information; or
- (e) as may be furnished to a Party's Affiliates, partners, directors, officers, employees, agents, consultants, auditors, banks, attorneys, or advisors who are required to keep the information in confidence

Notwithstanding the foregoing, a Party may disclose any one or more of the commercial terms of a Transaction to any industry price source for the purpose of aggregating and reporting such information in the form of a published energy price index, provided that such information does not include the identity of the other Party to a Transaction.

With respect to information provided with respect to a Transaction, this obligation shall survive for a period of one (1) year following the expiration or termination of such Transaction. With respect to information provided with respect to this Agreement, this obligation shall survive for a period of one (1) year following the termination of this Agreement.

21. PRIOR TRANSACTIONS

Any transaction entered into between the parties, now existing or hereafter, identified as a Transaction in this Agreement or the relevant Confirmation, whether before, on or after the effective date of this Agreement, is incorporated into this Agreement by reference, shall be a Transaction hereunder and shall be subject to the terms herein

22. TERMINATION

22.1 This Agreement may be terminated by either Party upon thirty (30) days' prior written notice of a Party's decision to terminate; provided however, that this Agreement shall remain in effect with respect to any Transactions entered into prior to such termination. The obligations of either Party to make payment hereunder or with respect to any Transactions entered into prior to the effective date of such termination, including any related adjustments, shall survive the termination of this Agreement.

EXHIBIT 1 SAMPLE FORM OF CONFIRMATION FOR LCFS CREDITS TRANSACTION

10.					
Attention:	***				
Phone Number:	***				
Facsimile Number:	***				
Email Address:	***				
Date:	***				
Reference:	***				
This Confirmation evide	ences the terms of the binding ag	greement between Seller and	Buyer named below reg	arding the Transaction.	
as of, a	lements, forms a part of, and is s s amended and supplemented b ns of the Agreement shall govern	by agreement in writing from	time to time (the "LEAP	LCFS Agreement") be	
	onsistency between this Confirm d terms used in this Confirmation t.				
TRADE DATE: ***					
SELLER: [Name and A	ddress]				
SELLER'S FEIN NUMI	BER:				
BUYER: [Name and Ad	ldress]				
BUYER'S FEIN NUMB	ER				
QUANTITY: *** LCFS	Credits.				
TRANSFER PERIOD:	***				
CONTRACT PRICE: U	JS\$ ***/ LCFS Credit				
CONTRACT AMOUNT	: US\$ ***				

PAYMENT DUE DATE: ***

SPECIAL CONDITIONS:

EXHIBIT 2 SAMPLE FORM OF INVOICE FOR LCFS CREDITS TRANSACTION

To: ***
Attention: ***
Phone Number: ***
Facsimile Number: ***
Email Address: ***
Date: ***
Invoice Number ***

TRADE DATE: ***

SELLER: [Name and Address]

SELLER'S FEIN NUMBER: BUYER: [Name and Address]

BUYER'S FEIN NUMBER: QUANTITY: *** LCFS Credits.

TRANSFER DATE: ***

CONTRACT PRICE: ***

CONTRACT AMOUNT: ***

SELLER BANKING DETAILS: ***

PAYMENT DUE DATE: ***

BOARD of SUPERVISORS



City Hall
Dr. Carlton B. Goodlett Place, Room 244
San Francisco 94102-4689
Tel. No. 554-5184
Fax No. 554-5163
TDD/TTY No. 554-5227

MEMORANDUM

TO:

Ben Rosenfield, City Controller, Office of the Controller

Ed Reiskin, Executive Director, Municipal Transportation Agency Harlan Kelly, Jr., General Manager, Public Utilities Commission

FROM:

Linda Wong, Assistant Clerk, Budget and Finance Sub-Committee

Board of Supervisors

DATE:

July 8, 2019

SUBJECT:

LEGISLATION INTRODUCED

The Board of Supervisors' Budget and Finance Sub-Committee has received the following proposed legislation:

File No. 190656

Ordinance delegating authority under Charter, Section 9.118(a), to the General Manager of the Public Utilities Commission to enter into one or more agreements to sell Low Carbon Fuel Standard Credits accrued by the Municipal Transportation Agency when revenue from such an agreement will be \$1,000,000 or more; and amending the Administrative Code to establish a Low Carbon Fuel Standard Credits Sales Fund to receive funds from the credits sales for use in connection with projects that will reduce San Francisco's carbon footprint and/or advance its sustainability goals.

If you have comments or reports to be included with the file, please forward them to me at the Board of Supervisors, City Hall, Room 244, 1 Dr. Carlton B. Goodlett Place, San Francisco, CA 94102 or by email at: linda.wong@sfgov.org.

c: Todd Rydstrom, Office of the Controller Kate Breen, Municipal Transportation Agency Janet Martinsen, Municipal Transportation Agency Joel Ramos, Municipal Transportation Agency John Scarpulla, Public Utilities Commission Christopher Whitmore, Public Utilities Commission

Office of the Mayor san francisco



LONDON N. BREED MAYOR

BOARD OF SUPERVISORS
SAM FRANCISCO

2019 JUN - 4 PM 3:56

y de

TO:

Angela Calvillo, Clerk of the Board of Supervisors

FROM:

Andres Power

RE:

Administrative Code - Granting Authority to Public Utilities Commission

General Manager to Sell Low Carbon Fuel Standard Credits, with Proceeds Used to Reduce City's Carbon Footprint and Advance

Sustainability Goals

DATE:

Tuesday, June 4, 2019

Ordinance delegating authority under Charter Section 9.118(a) to the General Manager of the San Francisco Public Utilities Commission to enter into one or more agreements to sell Low Carbon Fuel Standard Credits accrued by the San Francisco Municipal Transportation Agency when revenue from such an agreement will be \$1,000,000 or more; and amending the Administrative Code to establish a Low Carbon Fuel Standard Credits Sales Fund to receive funds from the credits sales for use in connection with projects that will reduce San Francisco's carbon footprint and/or advance its sustainability goals.

Should you have any questions, please contact Sophia Kittler at 415-554-6153.